CLASS ACTION SETTLEMENT AGREEMENT

This CLASS ACTION SETTLEMENT AGREEMENT ("Settlement Agreement") is entered into by and between Named Plaintiffs (as defined below) in the action titled In re Xerox Corporation ERISA Litigation, Civ. No. 3:02-CV-01128 (AWT) (D. Conn.) for themselves and on behalf of the Settlement Class (as defined below) and the Plans (as defined below), on the one hand, and the *Defendants* (as defined below) on the other, in consideration of the promises, covenants and agreements herein described and for other good and valuable consideration acknowledged by each of them to be satisfactory and adequate.

NOW, THEREFORE, without any admission or concession on the part of the Named Plaintiffs of any lack of merit of the action, and without any admission or concession on the part of Defendants as to the merit of the action, it is hereby STIPULATED AND AGREED, by and among the Parties (as defined below) to this Settlement Agreement, through their respective attorneys, subject to approval of the Court pursuant to the Federal Rules of Civil Procedure, in consideration of the benefits flowing to the Parties hereto from the Settlement Agreement, that all Released Claims (as defined below) as against the Released Parties (as defined below) shall be compromised, settled, released and dismissed with prejudice, upon and subject to the following terms and conditions:

1. Definitions.

As used in this Settlement Agreement, italicized and capitalized terms and phrases not otherwise defined herein have the meanings provided below:

- "Action" shall mean: In re Xerox Corporation ERISA Litigation, Civ. No. 3:02-CV-01128 (AWT), United States District Court for the District of Connecticut, and any and all cases now or hereafter consolidated therewith.
- "Agreement Execution Date" shall mean: the date on which this Settlement Agreement is fully executed, as provided in Section 12.11 below.
- "Other Appointed Counsel" shall mean: McTigue & Porter LLP, Stember 1.3 Feinstein Doyle & Payne, LLC, Barroway Topaz Kessler Meltzer Check, LLP, Berger & Montague, P.C., Law Offices of Daniel M. Harris, and Goodman Rosenthal & McKenna PC. and their predecessors and Successors-In-Interest.
 - 1.4 "Claims" shall have the meaning set forth in Section 3.2.
- "Class Notice" shall mean: the forms of notice appended as Exhibits 2 and 3 to the form of Preliminary Approval Order, attached hereto as Exhibit A.
 - 1.6 "Class Period" shall mean: May 12, 1997 through and including June 28, 2002.

- 1.7 "Co-Lead Counsel" shall mean: Lynn Sarko, Keller Rohrback L.L.P. and Charles R. Watkins, Futterman Howard Watkins Wylie & Ashley, Chtd. (previously of Susman & Watkins and Susman, Watkins & Wylie).
- "Company" or "Xerox" shall mean: Xerox Corporation, each of Xerox's 1.8 predecessors and Successors-In-Interest; any Person that controls, is controlled by, or is under common control with *Xerox*; any of its direct and indirect parents and subsidiaries, and any company whose employees participated in either of the Plans. For the purposes of this Settlement Agreement only, one company "controls" another company if (i) the former company, directly or indirectly, or acting through one or more other persons owns, has owned, or has or had the power to vote more than 50 percent of any class of voting securities of the latter company; or (ii) the former company has or had the right to determine in any manner the election of a majority of the directors or trustees of the latter company.
- "Complaint" shall mean: the Third Consolidated Amended Complaint in the Action, filed July 1, 2008.
- "Court" shall mean: the United States District Court for the District of 1.10 Connecticut.
- "Defendants" shall mean the following persons and/or entities: Lawrence Becker, 1.11 Sally Conkright, Patricia M. Nazemetz, Arlyn B. Kaster, Myra R. Drucker, Kathleen Russell, William Strusz, Lance Davis, Gregory B. Tayler, Christina Clayton, Gary Kabureck, Lawrence Zimmerman, Paul A. Allaire, Anne M. Mulcahy, William F. Buehler, Barry D. Romeril, B.R. Inman, Vernon E. Jordan, Jr., Hilmar Kopper, George J. Mitchell, N.J. Nicholas, Jr., Patricia F. Russo, Martha R. Seger, Thomas C. Theobald, G. Richard Thoman, Barbara D. Roscoe as beneficiary of property passing pursuant to the Last Will and Testament of William C. Roscoe, dated December 30, 2005, Henry Charles Filter, III, as beneficiary of property passing pursuant to the Will of Eunice M. Filter dated February 25, 2002 to the Eunice M. Filter Revocable Trust dated February 25, 2002, Henry Charles Filter, III, and John Musicaro, Jr., as Trustees of the Trust for the benefit of Henry Charles Filter, III and his descendants under Article II of the Eunice M. Filter Insurance Trust under Agreement dated January 19, 1995, as beneficiary of property passing pursuant to the Will of Eunice M. Filter dated February 25, 2002 to the Eunice M. Filter Revocable Trust dated February 25, 2002, c/o John Musicaro, Jr., Cummings & Lockwood LLC, Six Landmark Square, Stamford, CT 06901, and Jerry W. Hostetter and John Musicaro, Jr. as Trustees of the Marital Trust under Article IV of the Eunice M. Filter Revocable Trust dated February 25, 2002, as beneficiary of property passing pursuant to the Will of Eunice M. Filter dated February 25, 2002 to the Eunice M. Filter Revocable Trust dated February 25, 2002, c/o John Musicaro, Jr., Cummings & Lockwood LLC, Six Landmark Square, Stamford, CT 06901 (collectively, the "Individual Defendants"), and Xerox.
- "ERISA" shall mean: the Employee Retirement Income Security Act of 1974, as amended, including all regulations promulgated thereunder, and court decisions interpreting such Act or regulations.
 - "Fairness Hearing" shall have the meaning set forth in Section 2.2.4. 1.13

- 1.14 "Final" shall mean: with respect to any judicial ruling or order, that the period for any appeals, petitions, motions for reconsideration, rehearing, or certiorari or any other proceedings for review ("Review Proceeding") has expired without the initiation of a Review Proceeding, or, if a Review Proceeding has been timely initiated, that there has occurred a full and final disposition of any such Review Proceeding without a reversal or any material modification, including the exhaustion of proceedings in any remand and/or subsequent appeal after remand.
 - 1.15 "Financial Institution" shall have the meaning set forth in Section 7.1.1.
- 1.16 "Independent Fiduciary" shall mean the Person retained for the purposes set forth in Section 2.4.
- 1.17 "Judgment" shall have the meaning set forth in Section 2.2.4. A proposed form of the Judgment is attached hereto as Exhibit B.
- 1.18 "Named Plaintiffs" shall mean: Plaintiffs David Alliet, Linda Willis, Cheryl Wright and Thomas Patti.
 - 1.19 "Parties" shall mean: the Plaintiffs and the Defendants.
- 1.20 "Person" shall mean: an individual, partnership, corporation, governmental entity or any other form of entity or organization.
- 1.21 "Plaintiffs" shall mean: Named Plaintiffs and each member of the Settlement Class.
 - 1.22 "Plaintiff Releasees" shall have the meaning set forth in Section 3.3.
- 1.23 "Plans" shall mean: The Xerox Corporation Savings Plan and The Savings Plan of Xerox Corporation and the Xerographic Division, UNITE HERE (Formerly "The Savings Plan of Xerox Corporation and the Xerographic Division, Union of Needletrades, Industrial and Textile Employees, A.F.L.-C.I.O.-C.L.C."), together with their predecessors and successors, and any trust created under such plans.
- 1.24 "*Plan of Allocation*" shall mean: the plan of allocation approved by the Court. Attached as Exhibit C hereto is the form of plan of allocation that will be submitted to the Court for approval.
 - 1.25 "Preliminary Approval Order" shall have the meaning set forth in Section 2.2.1.
 - 1.26 "Preliminary Motion" shall have the meaning set forth in Section 2.2.1.
 - 1.27 "Released Claims" shall have the meaning set forth in Section 3.2.

- "Released Parties" shall mean: the Defendants, the Plans, every Person who, at any time during or after the Class Period, was a director, officer, employee or agent of the Company or a trustee or fiduciary of either or both of the Plans, together with, for each of the foregoing, any predecessors, Successors-In-Interest, present and former Representatives, direct or indirect parents and subsidiaries, insurers and any *Person* that controls, is controlled by, or is under common control with any of the foregoing.
 - 1.29 "Releases" shall mean the releases set forth in Article 3.
- 1.30 "Representatives" shall mean: attorneys, agents, directors, officers, and employees.
- "Securities Actions" shall mean: Carlson v. Xerox Corporation, Case No. 3:00-1.31 CV-1621 (AWT) (D. Conn.), In re: Xerox Securities Litig., 3:99-CV-02374 (AWT) (D. Conn.), and any and all cases now or hereafter consolidated therewith, and the SEC Fair Fund. Securities Actions shall not include the Action.
- "Settlement" shall mean: the settlement to be consummated under this Settlement 1.32 Agreement.
- "Settlement Administrator" shall mean: the Person who is so appointed pursuant 1.33 to the *Preliminary Approval Order* and who is responsible for calculating, implementing, and overseeing the Plan of Allocation.
- "Settlement Class" shall mean: all current and former participants and 1.34 beneficiaries of the Plans for whose individual accounts the Plans purchased and/or held interests in the Xerox Stock Fund at any time during the period May 12, 1997 through and including June 28, 2002, excluding the *Individual Defendants*.
 - "Settlement Fund" shall have the meaning set forth in Section 7.1. 1.35
- "Successor-In-Interest" shall mean: a Person's estate, legal representatives, heirs, successors or assigns, including successors or assigns that result from corporate mergers or other structural changes.

2. Implementation of the Settlement.

This Settlement shall be implemented in accordance with the provisions of this Settlement Agreement including Section 2 hereof, and, subject to the provisions of Section 10 hereof, shall be final when the conditions specified in Section 2 have been satisfied or waived.

2 1 Class Certification. The Court shall have certified this Action as a class action for settlement purposes only pursuant to Rule 23(a)(1)-(4), 23(b)(1)(A) and/or (B) and 23(e) of the Federal Rules of Civil Procedure, and with a Settlement Class defined as set forth in Section 1.34 above as a non-opt-out class. The Parties stipulate to certification of the Settlement Class for purposes of this Settlement only. The Parties agree that if the Court does not enter the

Judgment, or if the Judgment does not become Final, no Settlement Class will be deemed to have been certified by or as a result of this Settlement Agreement, and the Action will for all purposes with respect to the Parties revert to its status as of December 15, 2008. In such event the Defendants will not be deemed to have consented to the certification of any class, the agreements and stipulations in this Settlement Agreement concerning class definition or class certification shall not be used as evidence or argument to support class certification or class definition, and the *Defendants* will retain all rights to oppose class certification.

- 2.2 <u>Court Approval</u>. The *Settlement* shall have been approved by the *Court*, as provided for in this Section 2.2. Co-Lead Counsel shall move the Court for an order and judgment approving this Settlement Agreement and the Settlement contemplated hereunder in the form of the Judgment attached as Exhibit B hereto. The Named Plaintiffs shall recommend to the Court that such order and judgment be entered; the Defendants will not object to such recommendation; and the Parties shall cooperate in good faith, including taking all steps and efforts contemplated by this Settlement Agreement and any other steps or efforts that may become necessary by order of the Court (unless such order materially modifies the terms of this Settlement Agreement), to carry out this Settlement Agreement, including the following:
- 2.2.1 Motion for Preliminary Approval of Settlement and of Notice. As soon as reasonably possible upon the full execution of this Settlement Agreement by the Parties, Named Plaintiffs will file a motion or other appropriate document(s) as directed by the Court ("Preliminary Motion") with the Court for an order substantially in the form annexed hereto as Exhibit A, including the exhibits thereto (the "Preliminary Approval Order"). The Parties waive oral argument on the Preliminary Motion.
- 2.2.2 Request by Court, Named Plaintiffs, or Defendants for Information. If the Court requests or orders Named Plaintiffs or Defendants to supply non-privileged information in their possession as part of the Court's review of the Settlement Agreement, the Named Plaintiffs and Defendants agree to promptly provide such information. If Named Plaintiffs deem it necessary for the *Defendants* to supply non-privileged information in their possession, and not otherwise available to the Named Plaintiffs, in order to respond to any timely filed objection or Court request/order, the *Defendants* agree to promptly provide such non-privileged information that has been reasonably requested, which shall be treated as Confidential under the Amended Consent Protective Order entered by the Court in this Action on August 1, 2006. If Defendants deem it necessary for the *Named Plaintiffs* to supply non-privileged information in their possession in order to respond to any inquiry from the *Independent Fiduciary* or the Department of Labor, the Named Plaintiffs agree to promptly provide such non-privileged information that has been reasonably requested, which shall be treated as Confidential under the Amended Consent Protective Order entered by the *Court* in this *Action* on August 1, 2006.
- 2.2.3 <u>Issuance of Class Notice</u>. On the date and in the manner set by the Court in its Preliminary Approval Order, the Plaintiffs shall cause the Class Notice to be transmitted in the form and manner approved by the Court as directed in the Preliminary Approval Order. The Parties agree, and the form of Preliminary Approval Order shall provide, that the last known addresses for the *Plaintiffs* in the possession of the *Plans*' current record-keeper will suffice for

all purposes in connection with this Settlement and the Plan of Allocation, including, without limitation, the mailing of the Class Notice.

- The Fairness Hearing. On or after the date set by the Court for the hearing (the "Fairness Hearing") the Court will determine: (i) whether to enter judgment finally approving the Settlement (which judgment is referred to herein as the "Judgment"); and (ii) the amount of legal fees, compensation, and expenses to be awarded to Co-Lead Counsel and Other Appointed Counsel, and to the Named Plaintiffs as provided in Section 11 of this Settlement Agreement.
- Finality of *Judgment*. The Court shall have entered the *Judgment* substantially in the form attached as Exhibit B hereto, as more fully discussed in Section 2.2, and the Judgment shall have become Final.
- Settlement Authorized by Independent Fiduciary. At least twenty (20) days prior 2.4 to the Fairness Hearing, an Independent Fiduciary chosen by the Company, who shall acknowledge in writing that it is a fiduciary on behalf of the *Plans* with respect to the settlement of this Action, shall have approved and authorized in writing the Settlement in accordance with Prohibited Transaction Class Exemption 2003-39. If the *Independent Fiduciary* disapproves or otherwise does not authorize the Settlement, the Defendants shall have the option to waive this condition. Such option is to be exercised in writing within the earlier of (i) ten (10) days of the Defendants' receipt of the Independent Fiduciary's written determination, or (ii) three (3) days prior to the date set for the Fairness Hearing. If the Defendants elect in writing to waive this condition, the Settlement shall become Final if all other conditions in Section 2 are satisfied. If the Defendants elect not to waive this condition, the Settlement Agreement shall terminate and become null and void and the provisions of Section 10.2 shall apply.
- Assurances From the Department of Labor. Defendants may discuss this Settlement and the subject matter of this Action with the Department of Labor. If the Defendants discuss the Settlement or the subject matter of this Action with the Department of Labor and the Defendants fail to obtain reasonable assurance from the Department of Labor that it will not object to this Settlement or institute suit or an administrative proceeding against any Defendant at least twenty-five (25) days prior to the Fairness Hearing, and if Xerox, in its sole discretion, fails to waive this condition, then, upon Xerox notifying Named Plaintiffs in writing that this condition has failed, the Settlement Agreement shall terminate and become null and void and the provisions of Section 10.2 shall apply; provided that this reasonable assurance may be contingent on this Settlement becoming Final.
- Agreement With Settlement Administrator. Prior to the entry of the Preliminary 2.6 Approval Order, there shall be negotiated with the putative Settlement Administrator the form of agreement, acceptable to Named Plaintiffs and Xerox in their respective reasonable discretion, under which the Settlement Administrator will, subject to its appointment in the Preliminary Approval Order, perform the functions set forth in this Settlement Agreement relating to implementation of the *Plan of Allocation*. If such form of agreement is not reached prior to entry of the Preliminary Approval Order, then each of Named Plaintiffs and Xerox shall have the right to terminate this Settlement Agreement by written notice to the other, in which event the

Settlement Agreement shall terminate and become null and void and the provisions of Section 10.2 shall apply.

3. Releases.

- Releases of the Released Parties. Subject to Section 10 hereof, and the Class Settlement Amount stated in Paragraph 7.2 hereof having been deposited by Defendants into the Settlement Fund, effective upon the entry of the Judgment by the Court, Named Plaintiffs on behalf of themselves and on behalf of the Settlement Class and the Plans absolutely and unconditionally release and forever discharge with prejudice the Released Parties from Released Claims that the Plans, the Plaintiffs, the Settlement Class or Named Plaintiffs directly, indirectly, derivatively, or in any other capacity ever had, now have or hereafter may have, except that the release under this Section 3.1 shall not include claims relating to the covenants or obligations set forth in this Settlement Agreement. Also effective upon entry of the Judgment by the Court, the Plans, the Plans' fiduciaries, the Named Plaintiffs and all other members of the Settlement Class shall be permanently and finally enjoined without the necessity of posting a bond from commencing or prosecuting any actions or other proceedings asserting any of the Released Claims either directly, indirectly, derivatively, or in any other capacity, against any of the Released Parties.
- Released Claims. Subject to Section 3.4 below, the Released Claims shall be: any 3.2 and all claims of any nature whatsoever (including claims for any and all losses, damages, unjust enrichment, attorneys' fees, disgorgement of fees, litigation costs, injunction, declaration, contribution, indemnification or any other type or nature of legal or equitable relief), whether accrued or not, whether known, unknown, or unsuspected, in law or equity, as well as any claim or right obtained by assignment, brought by way of demand, complaint, cross-claim, counterclaim, third-party claim or otherwise (collectively, "Claims"), in any court or other tribunal, arising out of or in any way related to, directly or indirectly, any or all of the acts, omissions, facts, matters, transactions or occurrences, (i) that are, were or could have been alleged, asserted, or set forth in the Complaint or (ii) that would be barred by principles of res judicata had the claims asserted in the Complaint been fully litigated and resulted in a Final judgment or order. With respect to the *Released Claims*, it is the intention of the *Parties* and of the *Plans*, the *Named Plaintiffs* and all other members of the *Settlement Class* expressly to waive to the fullest extent of the law: (a) the provisions, rights and benefits of Section 1542 of the California Civil Code, which provides that "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor"; and (b) the provisions, rights and benefits of any similar statute or common law of any other jurisdiction that may be, or may be asserted to be, applicable. The Released Claims do not include any ERISA Section 502(a)(1)(B) claim for vested benefits by any Plaintiff where such claims are unrelated to any matter asserted in this Action.
- 3.3 <u>Defendants' Releases of Named Plaintiffs</u>, the <u>Settlement Class</u> and <u>Co-Lead and Other Appointed Counsel</u>. Subject to Section 10 hereof, effective upon the entry of the <u>Judgment</u> by the <u>Court</u>, the <u>Defendants</u> absolutely and unconditionally release and forever discharge with prejudice the <u>Named Plaintiffs</u>, the <u>Settlement Class</u>, <u>Co-Lead Counsel</u>, <u>Other</u>

Appointed Counsel, Dalton Gotto Samson & Kilgard PLC and Susman, Heffner & Hurst, LLP, and their and Co-Lead Counsel and Other Appointed Counsel's current and former members, partners, partnerships, professional corporations and firms, and their respective shareholders, partners, officers, directors, associates, agents, employees, consultants, Of Counsels, experts, vendors, predecessors (including all firms with which any Person currently employed or affiliated with Co-Lead Counsel or Other Appointed Counsel was employed or affiliated during the pendency of the Action, and the current and former members, partners, partnerships, shareholders, partners, officers, directors, associates, agents, employees, consultants, and Of Counsels of such firms), successors and assigns (collectively, the "Plaintiff Releasees"), from any and all *Claims* relating to the institution or prosecution of the *Action* or the settlement of any Released Claims, except that the release under this Section 3.3 shall not include claims relating to the covenants or obligations set forth in this Settlement Agreement or in the Amended Consent Protective Order entered by the *Court* in this *Action* on August 1, 2006.

<u>Claims Not Affected.</u> Nothing in this Settlement Agreement shall affect the settlement and release of claims in the Securities Actions, or bar, waive or otherwise affect any Claim that has been or could be asserted under state or federal securities laws by the Plans and/or any member of the Settlement Class in the Securities Actions.

4. Covenants.

The *Parties* covenant and agree as follows:

- Taxation of Class Settlement Amount. The Released Parties shall have no 4.1 responsibility for any taxes due with respect to funds deposited in, income earned on or funds distributed from the Settlement Fund, including funds that the Plaintiffs or Co-Lead Counsel or Other Appointed Counsel receive from the Class Settlement Amount.
- Cooperation. The Company shall cooperate with Co-Lead Counsel by providing, 4.2 in electronic format, the names and addresses of *Persons* to whom the *Class Notice* is to be sent pursuant to the Preliminary Approval Order, in accordance with Section 2.2.3. Plaintiffs shall use any information provided by the *Company* pursuant to the preceding sentence solely for the purpose of providing notice in this *Action* and for no other purpose. The *Parties* shall reasonably cooperate with each other to effectuate this Settlement, and shall not do anything or take any position inconsistent with obtaining a prompt Judgment approving the Settlement unless expressly permitted by this Settlement Agreement.
- Covenant Not to Sue. Subject to Section 10 hereof, *Plaintiffs* covenant and agree 4.3 on their own behalf, and on behalf of the Settlement Class and the Plans: (i) not to file against any Released Party any Claim based on, related to, or arising from any Released Claim; and (ii) that the foregoing covenants, agreements and releases shall be a complete defense to any such Claims against any of the respective Released Parties.
- Confidentiality. Except as otherwise provided by law or as may result in the ordinary course from submission of the Settlement to the Court pursuant to Section 2.2.1 hereof, the Parties agree that the Settlement shall remain confidential until the Company publicly

announces the Settlement. Co-Lead Counsel and Other Appointed Counsel agree to refrain from publishing, releasing, or otherwise publicly communicating any terms of the Settlement in advance of the public announcement of the Settlement by the Company without the prior approval of the Company acting on behalf of Defendants. Defendants retain the right to disclose the existence and terms of this *Settlement* to their external auditors and insurers at any time. Named Plaintiffs, on the one hand, and Defendants, on the other, further agree that they shall not disparage the other in any public statements regarding the Action, the Settlement, or the facts the subject matter of the Action. As used herein, "public statement" means any statement made in the press, broadcast, cable, or electronic media, or otherwise made in a manner designed to make the statement available to the public generally.

5. Representations and Warranties.

5.1 Named Plaintiffs' Representations and Warranties.

- 5.1.1 Named Plaintiffs represent and warrant that they have not assigned or otherwise transferred any interest in any Released Claims against any Released Party, and further covenant that they will not assign or otherwise transfer any interest in any Released Claims: and
- 5.1.2 Pursuant to Sections 3 and 4, *Named Plaintiffs* represent and warrant that upon this Settlement becoming final, they shall have no Released Claims against any of the Released Parties.
- Parties' Representations and Warranties. The Parties, and each of them, 5.2 represent and warrant as follows, each acknowledging that each is relying on these representations and warranties in entering into this Settlement Agreement:
- 5.2.1 That they have conducted voluminous discovery and have diligently prepared for trial pursuant to the Court's orders; that they are voluntarily entering into this Settlement Agreement as a result of arm's-length negotiations among their counsel, with the assistance of an experienced mediator; that in executing this Settlement Agreement they are relying solely upon their own judgment, belief and knowledge, and the advice and recommendations of their own independently selected counsel concerning the nature, extent and duration of their rights and claims hereunder and regarding all matters that relate in any way to the subject matter hereof; and that, except as otherwise explicitly set forth in this Settlement Agreement, they have not been influenced to any extent whatsoever in executing this Settlement Agreement by any representations, statements, or omissions pertaining to any of the foregoing matters by any Party or by any Person representing any Party to this Settlement Agreement. Each *Party* assumes the risk of mistake as to facts or law, except for the express representations and warranties set forth in this Settlement Agreement.
- 5.2.2 That they have carefully read the contents of this Settlement Agreement, and this Settlement Agreement is signed freely by each Person executing this Settlement Agreement on behalf of each of the Parties. The Parties, and each of them, further represent and warrant to each other that he, she, or it has made such investigation of the facts pertaining to the

Settlement, this Settlement Agreement, and all of the matters pertaining thereto, as he, she, or it deems necessary.

- 5.2.3 That in entering into this Settlement Agreement, the Plaintiffs have not relied on any representations or arguments by the Defendants that a class can, could, or should be certified for any purpose; that the Plaintiffs would not be prejudiced if (i) this Settlement were not approved or such approval were reversed on appeal and (ii) the Defendants were later to object to the certification of any proposed class in this action; and that in the event that this Settlement does not become Final, the Plaintiffs will not assert that the Defendants are equitably or judicially estopped from contesting the certification of any class in this Action.
- 5.3 <u>Signatories' Representations and Warranties</u>. Each *Person* executing this *Settlement Agreement* on behalf of any other *Person* does hereby personally represent and warrant to the other *Parties* that he or she has the authority to execute this *Settlement Agreement* on behalf of, and fully bind, each principal whom such individual represents or purports to represent.

6. No Admission of Liability.

The Parties understand and agree that this Settlement Agreement embodies a compromise settlement of disputed claims, and that nothing in this Settlement Agreement, including the furnishing of consideration for this Settlement Agreement, shall be deemed to constitute any finding of fiduciary status under ERISA or wrongdoing by any of the Defendants, or give rise to any inference of fiduciary status under ERISA or wrongdoing or admission of wrongdoing or liability in this or any other proceeding. This Settlement Agreement and the payments made hereunder are made in compromise of disputed claims and are not admissions of any liability of any kind, whether legal or factual. The Defendants specifically deny any such liability or wrongdoing and state that they are entering into this Settlement Agreement to eliminate the burden and expense of further litigation. Further, the Named Plaintiffs, while believing that all Claims brought in the Action have merit, have concluded that the terms of this Settlement Agreement are fair, reasonable and more than adequate to the Plans, themselves and members of the Settlement Class given, among other things, the inherent risks, difficulties and delays in complex ERISA litigation such as this. Neither the fact nor the terms of this Settlement Agreement shall be used or offered or received in evidence in any action or proceeding for any purpose, except in an action or proceeding to enforce this Settlement Agreement or arising out of or relating to the *Judgment*.

7. The Settlement Fund; Deliveries into the Settlement Fund.

7.1 The Settlement Fund.

7.1.1 Within five (5) days after the Agreement Execution Date, Co-Lead Counsel shall establish at a federally-insured financial institution (the "Financial Institution") identified to and reasonably agreed on by counsel for Xerox, a settlement fund account (the "Settlement Fund") which, along with net earnings thereon, shall be considered a common fund created as a result of the Action. Each Co-Lead Counsel firm and counsel for Xerox shall

designate at least one person with signature authority over this account (the "Signers"), and shall direct the Financial Institution to make distributions from the Settlement Fund only in accordance with this Settlement Agreement upon written direction from each Signer; provided that the Parties agree that upon the Judgment becoming Final, the Signers will notify the Financial Institution that written direction from counsel for Xerox shall no longer be required. For the avoidance of doubt, the Financial Institution shall be instructed that, absent a Court order, no funds are to be paid or withdrawn from the Settlement Fund except pursuant to Section 8 of this Settlement Agreement (and the Sections of this Settlement Agreement explicitly cross-referenced therein) or, upon termination of this Settlement Agreement, pursuant to Section 10 of this Settlement Agreement. Co-Lead Counsel shall promptly notify Xerox of the date of the establishment of the Settlement Fund, shall confirm the identity of the Financial Institution including any information, including but not limited to wiring instructions, needed to make the deposit in section 7.2, and shall confirm that withdrawals and distributions from the Settlement Fund are subject to the restrictions set forth in the preceding sentence.

7.1.2 The funds on deposit in the Settlement Fund shall be invested only in United States Treasury securities and/or securities of United States agencies backed by the full faith and credit of the United States Treasury, and mutual funds or money market accounts that invest exclusively in the foregoing securities. The Settlement Fund shall be structured and managed to qualify as a "qualified settlement fund" described in the Treasury regulations promulgated under Section 468B of the Internal Revenue Code and no Party shall take any position in any filing or before any tax authority that is inconsistent with such treatment. The Financial Institution or another person designated by Co-Lead Counsel (other than a Released Party) shall be the Settlement Fund "administrator," as that term is used in the Section 468B Treasury regulations (the "Administrator"). The Administrator shall (a) prepare and file all income tax and information returns required to be filed, and provide payees with copies of such information returns; (b) pay all taxes owed by the Settlement Fund; (c) at the Company's request, join with Xerox in timely making the "relation-back" election permitted under the Section 468B Treasury regulations in the form prescribed therein; (d) pay the fees and expenses incurred by the Financial Institution associated with the administration of the Settlement Fund and (e) obtain and provide the Company with the Settlement Fund's federal taxpayer identification number on or before the date that the Company transfers funds to the Settlement Fund. The Administrator shall be authorized to retain a certified public accounting firm for these purposes. All taxes on the income of the Settlement Fund and tax-related expenses, including the expenses, if any, of a certified public accounting firm, incurred in connection with the administration of the Settlement Fund shall be paid solely out of the Settlement Fund, shall be considered a cost of administration of the Settlement, and shall be timely paid without further order of the Court. All fees and expenses of the Administrator or the Financial Institution, and of professional advisors engaged by the Administrator or the Financial Institution in connection with the Settlement Fund, shall be funded solely from the Settlement Fund. Co-Lead Counsel, with written authority from the Signers, may instruct the Financial Institution to reserve any portion of the Settlement Fund for the purpose of satisfying future or contingent expenses or obligations, including expenses of Settlement Fund administration or any disbursement provided in Section 8 of this Settlement The Parties agree that neither Defendants nor any Released Party has any responsibility, authority, or liability, respecting the operation, expenses, obligations and administration of the Settlement Fund

- 7.2 The Class Settlement Amount. In consideration of, and expressly in exchange for, all of the promises and agreements set forth in this Settlement Agreement, the Company will cause to be deposited into the Settlement Fund within ten (10) business days after the entry of the Preliminary Approval Order or after the receipt of the wiring instructions and Financial Institution information referenced in Section 7.1.1, whichever is later, the aggregate sum of fifty-one million dollars in United States currency (\$51,000,000.00) (the "Class Settlement Amount").
- 7.3 <u>Sole Monetary Contribution</u>. The *Class Settlement Amount* shall be the full and sole monetary contribution made by or on behalf of the *Defendants* in connection with the *Settlement* effected between *Plaintiffs* and the *Defendants* under this *Settlement Agreement*. Provided the *Class Settlement Amount* is deposited in the *Settlement Fund* as provided in Section 7.1, no *Individual Defendant* will have any personal obligation to fund any or all of the *Class Settlement Amount*.

8. Payments From The Settlement Fund.

- 8.1 Expenses of Class Notice. After the entry of the Preliminary Approval Order, Co-Lead Counsel shall direct the Financial Institution in writing to disburse from the Settlement Fund an amount for the payment of reasonable, customary and feasible costs associated with the Class Notice, not to exceed \$200,000. If the Settlement Agreement is terminated for any reason, neither Co-Lead Counsel nor any other person shall have any obligation to reimburse to the Settlement Fund the costs of the Class Notice, or any other reasonable costs or expenses of the Settlement Fund charged to, or paid or incurred by or out of the Settlement Fund under this Settlement Agreement.
- 8.2 <u>Disbursements from Settlement Fund</u>. Except as provided in Section 7.1.2, Section 8.1, or in Section 10 below, no distribution of any part or all of the Settlement Fund shall be paid from the Settlement Fund until the Financial Institution has received (a) a joint notice signed by Co-Lead Counsel and by counsel for the Company, or (b) a Court order directing that the Settlement Fund be disbursed. When the condition in (a) or (b) has been satisfied, Co-Lead Counsel shall direct the Financial Institution to disburse money from the Settlement Fund as follows:
 - 8.2.1 <u>For Attorneys' Fees and Expenses</u>. As provided in Section 11.2 hereof.
 - 8.2.2 For *Named Plaintiff* compensation. As provided in Section 11.2 hereof.
- 8.2.3 <u>For taxes and expenses of the *Settlement Fund*</u>. As provided in Section 7.1.2 hereof.
- 8.2.4 For <u>Settlement Administrator Expenses</u>. Settlement Administrator Expenses shall mean all reasonable fees and expenses of the <u>Settlement Administrator</u> to calculate, implement, and oversee the <u>Plan of Allocation</u>, including, but not limited to, expenses incurred in calculating and allocating the share of the <u>Net Proceeds</u>, as defined in Section 8.2.5, for each member of the <u>Settlement Class</u>. The <u>Settlement Fund</u> (and this allocation) shall be the

12

sole source used to pay for these fees and expenses. The *Parties* agree to cooperate to seek to minimize the *Settlement Administrator Expenses*. The *Parties* also agree that *Settlement Administrator Expenses* shall not be incurred until reasonably necessary in light of the conditions to the *Settlement*.

8.2.5 For the *Plan of Allocation*. The *Plan of Allocation* attached as Exhibit C hereto provides for the allocation of the Settlement Fund net of the disbursements called for in Sections 8.2.1, 8.2.2, 8.2.3, and 8.2.4 ("Net Proceeds"). Upon the Judgment becoming Final as provided in Section 2.3, and after the amounts payable pursuant to Sections 8.2.1, 8.2.2, 8.2.3 and 8.2.4 have been determined and disbursed, Co-Lead Counsel shall direct the Financial Institution to disburse the Net Proceeds to the master trust for the Plans for allocation to or for the benefit of members of the Settlement Class. The Parties agree that the deposit of the Net Proceeds into the trust created under the Plans shall constitute "restorative payments" within the meaning of Revenue Ruling 2002-45 for all purposes. *Defendants* shall direct State Street Bank and Trust Company, or any successor Trustee of the *Plans*, or any other authorized entity, to allocate the Net Proceeds received by the Plans' trust according to the Plan of Allocation and shall notify Co-Lead Counsel as to the date(s) and amount(s) of said allocation(s), but Defendants shall have no liability for the failure by State Street Bank and Trust Company, or any successor Trustee of the *Plans*, the *Plans*' record-keeper, or any other authorized entity, to follow such directions provided that, in the event of any failure by State Street Bank and Trust Company, or any successor Trustee of the Plans, or any other authorized entity, to follow directions from Defendants given pursuant to this Section 8.2.5, the Defendants shall assist in seeking to enforce such directions. All reasonable fees and expenses of State Street Bank and Trust Company, or any predecessor or successor Trustee of the *Plans*, the *Plans*' record-keeper or any other authorized entity, with respect to implementation of the *Plan of Allocation* shall be Plans' expenses paid solely out of the Net Proceeds and shall be timely paid by the Plans without further order of the Court. Defendants warrant that they either have obtained or will obtain the authority to direct that the Net Proceeds received by the Plans' trust be allocated according to the Plan of Allocation. Defendants shall have no responsibility for structuring the content of the Plan of Allocation, or for its design or implementation, but will have the right to review it for feasibility and cost before presentation to the Court. The Plan of Allocation is a matter separate and apart from the Settlement between the Parties, and no decision by the Court that modifies the Plan of Allocation shall in any manner affect the validity of this Settlement Agreement, or the finality of the Settlement, or the binding character of the releases provided for hereunder.

9. Additional Conditions

9.1 <u>Asset Management</u>. The Xerox Retirement Investment Committee ("XRIC"), consisting of specified members of the *Company*'s management (Treasurer; Chief Accounting Officer; Vice President, Worldwide Taxes; Vice President, Human Resources; and Chief Information Officer), will abide by the terms of each of the *Plans* with respect to asset management, including asset management of the Xerox Stock Fund, in a manner consistent with *ERISA* and other applicable law.

- 9.2 Other Funds. Unless otherwise required by law, the *Company* shall not amend the *Plans* to prevent participants and beneficiaries from investing all or any part of their account balances in funds other than the Xerox Stock Fund.
- 9.3 <u>XRIC Composition</u>. The composition of *XRIC* will be identified to participants in the *Plans*.
- 9.4 <u>Disclosure</u>. Annually, the investment objectives of the Xerox Stock Fund will be published to the participants and beneficiaries of each of the *Plans*.
- 9.5 <u>Independent Fiduciary</u>. The *Company* shall pay the costs of the *Independent Fiduciary* using corporate assets.

10. <u>Termination of the Settlement Agreement.</u>

- 10.1 <u>Termination</u>. This *Settlement Agreement* may terminate if (a) the *Court* declines to enter the *Judgment*, (b) the *Judgment* entered by the *Court* is reversed or modified in any material respect on appeal, (c) the *Independent Fiduciary* does not approve the *Settlement* as set forth in Section 2.4, and such condition is not waived in writing by *Defendants*, or (d) the Department of Labor takes any of the actions described in Section 10.3. This *Settlement Agreement* shall not terminate if a court of competent jurisdiction modifies, reverses, or refuses to enter any order relating to the award of legal fees and expenses, or *Named Plaintiff* compensation, or the *Plan of Allocation*. If, within thirty (30) days after the date when any reversal or modification which would cause this *Settlement Agreement* to terminate becomes *Final*, the *Parties* have not agreed in writing to proceed with all or part of the *Settlement Agreement* in light of such ruling, this *Settlement Agreement* shall automatically terminate and thereupon become null and void.
- 10.2 <u>Consequences of Termination of the Settlement Agreement.</u> If the Settlement Agreement terminates, the following shall occur:
- 10.2.1 Co-Lead Counsel and Defendants' Counsel shall within ten (10) days after the date of termination of the Settlement Agreement jointly notify the Financial Institution in writing to return to the Defendants the amount contributed by them to the Settlement Fund, with all net income earned thereon, after deduction of the amount earlier disbursed for Class Notice described in Section 8.1 and the expenses charged by the Administrator and the Financial Institution, directing the Financial Institution to effect such return within fourteen (14) days after such notification. Prior to the return of amounts contemplated by this Section 10.2.1, the Administrator shall fully and finally fulfill all tax obligations of the Settlement Fund as set forth in Section 7.1.2 and the Defendants shall have no past, present, or future liability whatsoever for any such tax obligations.
- 10.2.2 The *Action* shall for all purposes with respect to the *Parties* revert to its status as of December 15, 2008. Any and all statutes of limitations, statutes of repose and/or other defenses based upon the passage of time applicable to the *Claims* asserted in this *Action*

shall be tolled from December 15, 2008 to the date on which this Settlement Agreement terminates

- 10.2.3 All provisions of this *Settlement Agreement* shall be null and void except as otherwise explicitly provided in this *Settlement Agreement*.
- 10.3 <u>Intervention by the Department of Labor</u>. If, prior to the entry of the *Judgment* in this case, the Department of Labor files an objection to the *Settlement*, files an amicus curiae brief disapproving of or opposing the *Parties' Settlement* in this *Action* or files its own lawsuit against any of the *Defendants* asserting allegations and claims similar to those asserted in the *Action, Xerox* shall have the right to terminate this *Settlement Agreement* at its sole discretion.

11. Attorneys' Fees and Expenses and *Named Plaintiff* Compensation.

- 11.1 <u>Application for Attorneys' Fees and Expenses and Named Plaintiff</u> <u>Compensation</u>. As provided in Section 2.2, and pursuant to the common fund doctrine and/or any applicable statutory fee provision, *Co-Lead Counsel* may apply to the *Court* for an award to *Co-Lead Counsel* and *Other Appointed Counsel* of attorneys' fees, and for reimbursement of expenses, to be paid solely from the *Settlement Fund*. *Co-Lead Counsel* also may apply to the *Court* for compensation to *Named Plaintiffs*, payable solely from the *Settlement Fund*, and *Named Plaintiffs* shall be entitled to receive such compensation from the *Settlement Fund* to the extent awarded by the *Court*.
- 11.2 <u>Disbursement of Attorneys' Fees and Expenses and Named Plaintiff</u> Compensation. If the Court enters one or more orders allowing payment of attorneys' fees and/or expenses and/or Named Plaintiff compensation, then Co-Lead Counsel may, not less than five (5) days after the Judgment becomes Final, instruct the Financial Institution in writing to disburse the payments set forth in such order(s) from the Settlement Fund.

12. Miscellaneous Provisions.

- 12.1 <u>Governing Law</u>. This *Settlement Agreement* shall be governed by the laws of the State of New York without giving effect to the conflict of laws or choice of law provisions thereof, except to the extent that the law of the United States governs any matter set forth herein, in which case such federal law shall govern.
- 12.2 <u>Amendment</u>. Before entry of the *Judgment*, the *Settlement Agreement* may be modified or amended only by written agreement signed by or on behalf of all *Parties*. Following entry of the *Judgment*, the *Settlement Agreement* may be modified or amended only by written agreement signed by or on behalf of all *Parties* and approved by the *Court*.
- 12.3 <u>Waiver</u>. The provisions of this *Settlement Agreement* may be waived only by an instrument in writing executed by the waiving party. The waiver by any party of any breach of this *Settlement Agreement* shall not be deemed to be or construed as a waiver of any other breach, whether prior, subsequent, or contemporaneous, of this *Settlement Agreement*.

- 12.4 <u>Construction</u>. None of the *Parties* hereto shall be considered to be the drafter of this *Settlement Agreement* or any provision hereof for the purpose of any statute, case law, or rule of interpretation or construction that would or might cause any provision to be construed against the drafter hereof.
- 12.5 <u>Principles of Interpretation</u>. The following principles of interpretation apply to this *Settlement Agreement:*
- 12.5.1 <u>Headings</u>. The headings of this *Settlement Agreement* are for reference purposes only and do not affect in any way the meaning or interpretation of this *Settlement Agreement*.
- 12.5.2 <u>Singular and Plural</u>. Definitions apply to the singular and plural forms of each term defined.
- 12.5.3 <u>Gender</u>. Definitions apply to the masculine, feminine, and neuter genders of each term defined.
- 12.5.4 <u>References to a Person</u>. References to a Person are also to the Person's permitted successors and assigns.
- 12.5.5 <u>Terms of Inclusion</u>. Whenever the words "include," "includes," or "including" are used in this *Settlement Agreement*, they shall not be limiting but rather shall be deemed to be followed by the words "without limitation."
- 12.5.6 <u>Time</u>. References to "days" in this *Settlement Agreement* are to calendar days, unless otherwise stated.
- 12.6 <u>Further Assurances</u>. All *Parties* agree, without further consideration, and as part of finalizing the *Settlement* hereunder, that they will in good faith execute and deliver such other documents and take such other actions as may be necessary to consummate and effectuate the subject matter and purpose of this *Settlement Agreement*.
- 12.7 <u>Notices</u>. Any notice, demand, or other communication under this *Settlement Agreement* (other than the *Class Notice*, or other notice given at the direction of the *Court*) shall be in writing and shall be deemed duly given upon receipt if it is addressed to each of the intended recipients as set forth below and personally delivered, sent by registered or certified mail (postage prepaid), sent by confirmed facsimile, or delivered by reputable express overnight courier:

IF TO NAMED PLAINTIFFS:

Lynn Sarko
Gary Gotto
Elizabeth Leland
Keller Rohrback L.L.P.

1201 Third Avenue, Suite 3200 Seattle, WA 98101 Telephone: (206) 623-1900

Facsimile: (206) 623-3384

Charles Watkins John Wylie Futterman Howard Watkins Wylie & Ashley, Chtd 122 S. Michigan Avenue, Suite 1850

Chicago, Illinois 60603 Telephone: (312) 427-3600 Facsimile: (312) 427-1850

IF TO DEFENDANTS

Steven Sacher Evan Miller Jones Day 51 Louisiana Ave. N.W. Washington, DC 20001-2113 Telephone: (202) 879-3939 Facsimile: (202) 626-1700

Any Party may change the address at which it is to receive notice by written notice delivered to the other *Parties* in the manner described above.

- Entire Agreement. This Settlement Agreement contains the entire agreement among the *Parties* relating to this *Settlement*. It specifically supersedes any settlement terms or settlement agreements relating to the *Defendants* that were previously agreed upon orally or in writing by any of the *Parties*. In the event of any dispute regarding the interpretation of this Settlement Agreement, no Party may rely upon or cite any previous version of this Settlement Agreement.
- Counterparts. This Settlement Agreement may be executed by exchange of faxed or scanned executed signature pages, and any signature thereby transmitted for the purpose of executing this Settlement Agreement shall be deemed an original signature for purposes of this Settlement Agreement. This Settlement Agreement may be executed in two or more counterparts, each of which shall be deemed to be an original, but all of which, taken together, shall constitute one and the same instrument.
- 12.10 Binding Effect. This Settlement Agreement binds and inures to the benefit of the Parties hereto, their assigns, heirs, administrators, executors, and successors.
- 12.11 Agreement Execution Date. The date on which the final signature is affixed below shall be the *Agreement Execution Date*.

Dated: 1/26/09

12.12 <u>Return of Discovery Documents</u>. Within sixty (60) days after the *Judgment* becomes *Final*, the parties shall comply with the provisions of the Amended Consent Protective Order, issued by the *Court* in this *Action* on August 1, 2006.

IN WITNESS WHEREOF, the Parties have executed this Settlement Agreement on the dates set forth below.

FOR THE PLAINTIFES:

By:

Lynn Sarko

Gary Gotto Elizabeth Leland

Keller Rohrback L.L.P.

1201 Third Avenue, Suite 3200

Seattle, WA 98101

Telephone: (206) 623-1900 Facsimile: (206) 623-3384

By:

Charles Watkins

John Wylie Futterman Howard Watkins Wylie & Ashley, Chtd

122 S. Michigan Avenue, Suite 1850

Chicago, Illinois 60603 Telephone: (312) 427-3600 Facsimile: (312) 427-1850

Co-Lead Counsel for the ERISA Plaintiffs

FOR DEFENDANTS XEROX CORPORATION, LAWRENCE BECKER, SALLY CONKRIGHT, PATRICIA M. NAZEMETZ, ARLYN B. KASTER, KATHLEEN RUSSELL, WILLIAM STRUSZ, LANCE DAVIS, CHRISTINA CLAYTON, GARY KABURECK, LAWRENCE ZIMMERMAN, ANNE M. MULCAHY, B.R. INMAN, VERNON E. JORDAN, JR., HILMAR KOPPER, GEORGE J. MITCHELL, N.J. NICHOLAS, JR., PATRICIA F. RUSSO, MARTHA R. SEGER, THOMAS C. THEOBALD, G. RICHARD THOMAN, BARBARA D. ROSCOE AS BENEFICIARY OF PROPERTY PASSING PURSUANT TO THE LAST WILL AND TESTAMENT OF WILLIAM C. ROSCOE, DATED DECEMBER 30, 2005, HENRY CHARLES FILTER, III. AS BENEFICIARY OF PROPERTY PASSING PURSUANT TO THE WILL OF EUNICE M. FILTER DATED FEBRUARY 25, 2002 TO THE EUNICE M. FILTER REVOCABLE TRUST DATED FEBRUARY 25. 2002, HENRY CHARLES FILTER, III, AND JOHN MUSICARO, JR., AS TRUSTEES OF THE TRUST FOR THE BENEFIT OF HENRY CHARLES FILTER, III AND HIS DESCENDANTS UNDER ARTICLE II OF THE EUNICE M. FILTER INSURANCE TRUST UNDER AGREEMENT DATED JANUARY 19, 1995, AS BENEFICIARY OF PROPERTY PASSING PURSUANT TO THE WILL OF EUNICE M. FILTER DATED FEBRUARY 25, 2002 TO THE EUNICE M. FILTER REVOCABLE TRUST DATED FEBRUARY 25, 2002, C/O JOHN MUSICARO, JR., CUMMINGS & LOCKWOOD LLC, SIX LANDMARK SQUARE, STAMFORD, CT 06901, AND JERRY W. HOSTETTER AND JOHN MUSICARO, JR. AS TRUSTEES OF THE MARITAL TRUST UNDER ARTICLE IV OF THE EUNICE M. FILTER REVOCABLE TRUST DATED FEBRUARY 25, 2002, AS BENEFICIARY OF PROPERTY PASSING PURSUANT TO THE WILL OF EUNICE M. FILTER DATED FEBRUARY 25, 2002 TO THE EUNICE M. FILTER REVOCABLE TRUST DATED FEBRUARY 25. 2002, C/O JOHN MUSICARO, JR., CUMMINGS & LOCKWOOD LLC. SIX LANDMARK SQUARE, STAMFORD, CT:

By:

Steven J. Sacher Evan Miller

Jones Day

51 Louisiana Ave. N.W.

Washington, DC 20001-2113 Telephone: (202) 879-3939

Facsimile: (202) 626-1700

Attorneys for the Defendants except Myra R. Drucker, Paul A. Allaire, William F. Buehler, Barry D. Romeril, and Gregory B. Tayler

FOR DEFENDANT MYRA R. DRUCKER:

By:	Blot T. Hander	Dated:	January	26,2009
25.	Michael T. Hannafari			*
	Blake T. Hannafan			
	Hannafan & Hannafan, Ltd.			
	One East Wacker Drive, Suite 2800			
	Chicago, IL 60601			
	Telephone: (312) 527-0055			
	Attorneys for Defendant Myra R. Dru	cker		
FOR	DEFENDANTS PAUL A. ALLAIRE, '	WILLLA	M F. BUEHLER	
	RY D. ROMERIL, AND GREGORY B			
By:		Dated:		
-	John A. Valentine			
	Wilmer Cutler Pickering Hale and Do	orr LLP		
	1875 Pennsylvania Ave., NW		•	
	Washington, DC 20006			
	Telephone: (202) 663-6000			
	Attorneys for Defendants Paul A. Allo		lliam F. Buehler,	
	Barry D. Romeril, and Gregory B. Ta	ıyler		
FOR	DEFENDANT G. RICHARD THOMA	N:		
D		Dated:		
By:	Thomas D. Goldberg	Date.		
	Day Pitney LLP			
	One Canterbury Green			
	Stamford, CT 06901			
	Telephone: (203) 977-7300			
	1616phone. (203) 777-7300			

Attorneys for Defendant G. Richard Thoman

Attorneys for the Defendants except Myra R. Drucker, Paul A. Allaire, William F. Buehler, Barry D. Romeril, and Gregory B. Tayler

FOR DEFENDANT MYRA R. DRUCKER:

By:	Dated:			
	Michael T. Hannafan Blake T. Hannafan Hannafan & Hannafan, Ltd. One East Wacker Drive, Suite 2800 Chicago, IL 60601 Telephone: (312) 527-0055			
	Attorneys for Defendant Myra R. Drucker			
	DEFENDANTS PAUL A. ALLAIRE, WILLIAM F. BUEHLER, Y D. ROMERIL, AND GREGORY B. TAYLER:			
By:	John A. Valentine Wilmer Cutler Pickering Hale and Dorr LLP 1875 Pennsylvania Ave., NW Washington, DC 20006 Telephone: (202) 663-6000 Attorneys for Defendants Paul A. Allaire, William F. Buehler, Barry D. Romeril, and Gregory B. Tayler			
FOR DEFENDANT G. RICHARD THOMAN:				
Ву:	Thomas D. Goldberg Day Pitney LLP One Canterbury Green Stamford, CT 06901 Telephone: (203) 977-7300			
	Attorneys for Defendant G. Richard Thoman			

Attorneys for the Defendants except Myra R. Drucker, Paul A. Allaire, William F. Buehler, Barry D. Romeril, and Gregory B. Tayler

FOR DEFENDAN	T	MYRA	R.	DRUCKER:
--------------	---	------	----	----------

_				
Ву:	Dated:			
Michael T.				
Blake T. H				
	Hannafan & Hannafan, Ltd.			
	Vacker Drive, Suite 2800			
Chicago, II				
Telephone:	(312) 527-0055			
Attorneys f	or Defendant Myra R. Drucker			
FOR DEFENDANTS PAUL A. ALLAIRE, WILLIAM F. BUEHLER, BARRY D. ROMERIL, AND GREGORY B. TAYLER:				
By:	Dated:			
John A. Va	The second of th			
Wilmer Cu	tler Pickering Hale and Dorr LLP			
	sylvania Ave., NW			
	n, DC 20006			
_	(202) 663-6000			
•				
Attorneys f	or Defendants Paul A. Allaire, William F. Buehler,			
	omeril, and Gregory B. Tayler			
	- · ·			
FOR DEFENDAN	T G. RICHARD THOMAN:			
Ву:	0 Old Dated: 1-22-09			
Thomas D.				
Day Pitney				
	bury Green			
Stamford, (
Telephone:	(203) 977-7300			
Attorneys fo	or Defendant G. Richard Thoman			

EXHIBITS TO THE SETTLEMENT AGREEMENT

Exhibits

A Preliminary Approval Order

B Judgment

C Plan of Allocation

EXHIBIT A TO CLASS ACTION SETTLEMENT AGREEMENT

(ORDER PRELIMINARILY APPROVING SETTLEMENT AND CONFIRMING FINAL SETTLEMENT HEARING)

UNITED STATES DISTRICT COURT DISTRICT OF CONNECTICUT

In re XEROX CORPORATION ERISA LITIGATION

This Document Relates To:

All Actions

Master File No. 02-CV-1138 (AWT)

CLASS ACTION

January 26, 2009

ORDER PRELIMINARILY APPROVING SETTLEMENT AND CONFIRMING FINAL SETTLEMENT HEARING

This litigation involves claims for alleged violations of the Employee Retirement Income Security Act of 1974, as amended, 29 U.S.C. §§ 1001, et seq. ("ERISA"), with respect to The Xerox Corporation Savings Plan and The Savings Plan of Xerox Corporation and the Xerographic Division, UNITE HERE (Formerly "The Savings Plan of Xerox Corporation and the Xerographic Division, Union of Needletrades, Industrial and Textile Employees, A.F.L.-C.I.O.-C.L.C."), together with their predecessors and successors, and any trust created under such plans (the "Plans").

Presented to the Court for preliminary approval is a settlement of the litigation as against all Defendants (the "Settlement"). The terms of the Settlement are set out in a Class Action Settlement Agreement (the "Settlement Agreement"), executed by counsel on January 26, 2009 on behalf of the Parties.

On January 26, 2009, Plaintiffs submitted the Settlement Agreement for preliminary approval. The Court has considered the Settlement to determine, among other things, whether to

_

Capitalized terms not otherwise defined in this order shall have the same meaning as ascribed to them in the Settlement Agreement.

certify a settlement class and whether the Settlement is sufficient to warrant the issuance of notice to members of the Settlement Class. Upon reviewing the Settlement Agreement, it is hereby ORDERED, ADJUDGED AND DECREED as follows:

- 1 Preliminary Findings Regarding Proposed Settlement. The Court finds on a preliminary basis that the Settlement Agreement filed and incorporated herein by reference, and made a part of this Order of Preliminary Approval, appears to be within the scope of reasonableness of a settlement which could ultimately be given final approval by this Court; it further appears to this Court on a preliminary basis, that the settlement amount is fair and reasonable to Class Members when balanced against the probable outcome of further litigation relating to liability and damages issues and potential appeals of rulings; that significant discovery, investigation, research, and litigation has been conducted such that counsel for the parties at this time are able to reasonably evaluate their respective positions; that settlement at this time will avoid substantial costs, delay and risks that would be presented by the further prosecution of the litigation; that the proposed Settlement resulted from informed, extensive noncollusive arm's-length negotiations and numerous mediation sessions over a period of seven months before Robert Meyer, and followed years of contested motion practice and discovery; and that the proposed Settlement is sufficiently fair, reasonable, and adequate to warrant sending notice of the Settlement to the Settlement Class.
- 2. <u>Class Findings.</u> The Court preliminarily finds that the requirements of the Federal Rules of Civil Procedure, the United States Constitution, the Rules of the Court and any other applicable law have been met as to the "Settlement Class" defined below, in that:
 - (a) The Court preliminarily finds that Rule 23(a)(1) is satisfied because the Settlement Class is ascertainable from records kept with respect to the Plans and from other objective criteria, and the members of the Settlement

- Class are so numerous that their joinder before the Court would be impracticable.
- (b) The Court preliminarily finds that Rule 23(a)(2) is satisfied because there are one or more questions of fact and/or law common to the Settlement Class.
- (c) The Court preliminarily finds that Rule 23(a)(3) is satisfied because the Named Plaintiffs' claims are typical of the claims of the Settlement Class.
- (d) The Court preliminarily finds that Rule 23(a)(4) is satisfied because the Named Plaintiffs will fairly and adequately protect the interests of the Settlement Class in that: (i) the Named Plaintiffs' interests and the nature of claims alleged are consistent with those of the members of the Settlement Class; (ii) there appear to be no conflicts between or among the Named Plaintiffs and the Settlement Class; and (iii) the Named Plaintiffs and the members of the Settlement Class are represented by qualified, reputable counsel who are experienced in preparing and prosecuting large, complicated ERISA class actions.
- (e) The Court preliminarily finds that Rule 23(b)(1)(A) and (B) are satisfied because the prosecution of separate actions by individual members of the Settlement Class would create a risk of: (i) inconsistent or varying adjudications as to individual class members that would establish incompatible standards of conduct for Defendants; or (ii) adjudications as to individual class members that would, as a practical matter, be dispositive of the interests of the other members not parties to the adjudications, or substantially impair or impede those persons' ability to protect their interests.
- (f) The Court preliminarily finds Rule 23(g) is satisfied because Keller Rohrback L.L.P. and Futterman Howard Watkins Wylie & Ashley, Chtd. ("Class Counsel") are capable of fairly and adequately representing the interests of the Settlement Class. Class Counsel have done extensive work identifying or investigating potential claims in the action, and have vigorously litigated the validity of those claims for the past six years. Class Counsel are experienced in handling class actions, other complex litigation, and claims of the type asserted in the Action. Class Counsel are knowledgeable of the applicable law, and Class Counsel have committed the necessary resources to represent the Settlement Class.

ACCORDINGLY, GOOD CAUSE APPEARING, PRELIMINARY APPROVAL OF SETTLEMENT IS HEREBY GRANTED; THE CLASS IS CONDITIONALLY CERTIFIED FOR SETTLEMENT PURPOSES ONLY.

3. <u>Class Certification</u>. Based on the findings set forth above, and pursuant to Fed. R. Civ. P. 23(a) and (b)(1), the Court conditionally certifies the following class for settlement purposes only (the "Settlement Class"):

all current and former participants and beneficiaries of the Plans for whose individual accounts the Plans purchased and/or held interests in the Xerox Stock Fund at any time during the period May 12, 1997 through and including June 28, 2002, excluding the *Individual Defendants*.

The Court appoints the Named Plaintiffs as the class representatives for the Settlement Class, and Keller Rohrback L.L.P., and Futterman Howard Watkins Wylie & Ashley, Chtd. as Class Counsel for the Settlement Class.

- 4. <u>Settlement Administrator</u>. The Court hereby appoints Gilardi & Co LLC as Settlement Administrator. The Settlement Administrator shall have the responsibilities stated in section 1.33 of the Settlement Agreement, as described more fully in the scope of work description attached hereto as Exhibit 1, and shall cause mailing and publication of the Class Notice as provided in Paragraph 6, below.
- 5. <u>Fairness Hearing</u>. A hearing is scheduled for ______, 2009 (the "Fairness Hearing") to determine, among other things:
 - (a) Whether the Settlement should be approved as fair, reasonable, and adequate;
 - (b) Whether the Complaint should be dismissed with prejudice pursuant to the terms of the Settlement:
 - (c) Whether the Class Notice provided for by the Settlement Agreement was provided and: (i) appropriate and reasonable and constituted due, adequate, and sufficient notice to all persons entitled to notice; and (ii) met all applicable requirements of the Federal Rules of Civil Procedure, and any other applicable law;
 - (d) Whether Class Counsel adequately represented the Settlement Class for purposes of entering into and implementing the Settlement Agreement;

- (e) Whether the application for attorneys' fees and expenses filed by Class Counsel should be approved; and
- (f) Whether the application for Named Plaintiff compensation awards for the Named Plaintiffs should be approved.
- 6. <u>Class Notice</u>. A proposed form of Class Notice is attached hereto as Exhibit 2. With respect to such form of Class Notice, the Court finds that such form fairly and adequately: (a) describes the terms and effect of the Settlement Agreement and of the Settlement; (b) notifies the Settlement Class concerning the proposed Plan of Allocation; (c) notifies the Settlement Class that Class Counsel will seek from the Settlement Fund awards of attorneys fees and expenses, reimbursement of expenses, and case contribution awards for the Named Plaintiffs; (d) gives notice to the Settlement Class of the time and place of the Fairness Hearing; and (e) describes how the recipients of the Class Notice may object to any of the relief requested. The Court directs that Co-Lead Counsel shall cause the Settlement Administrator to:
 - (a) By no later than forty-five (45) days before the Fairness Hearing, cause the Class Notice, with such non-substantive modifications thereto as may be agreed upon by the Parties, to be sent to each Person within the Settlement Class who can be identified by Defendants with reasonable effort. Such notice shall be sent either by e-mail or first-class mail, postage prepaid, to the Person's last known address as set forth in the records of Plans' record-keeper. The Company shall cause the Plans' record-keeper to provide to Class Counsel the names and last known addresses of the Class members in electronic format at least twenty-one (21) days prior to the deadline for mailing notice.
 - (b) By no later than forty-five (45) days before the Fairness Hearing, cause the Class Notice to be published on each website identified in the Class Notice.
 - (c) By no later than forty-five (45) days before the Fairness Hearing, cause a summary notice in the form attached hereto as Exhibit 3, with such non-substantive modifications thereto as may be agreed upon by the Parties, to be published on at least one occasion in the New York Times (National

Edition), Rochester Democrat & Chronicle, the Connecticut Post, the Norwalk Hour and the Business Wire.

At or before the Fairness Hearing, Class Counsel shall file with the Court a statement of timely compliance with the foregoing mailing and publication requirements.

- Objections to Settlement. Any member of the Settlement Class who wishes to object to the fairness, reasonableness, or adequacy of the Settlement, to the Plan of Allocation, to any term of the Settlement Agreement, to the proposed award of attorneys' fees and expenses, or to any request for a case contribution award for the Named Plaintiffs, may file an Objection in accordance with the procedures set forth in the Class Notice no later than fourteen (14) days prior to the Fairness Hearing.
- 8. Appearance at Fairness Hearing. Any objector who files and serves a timely, written objection in accordance with paragraph 6 above, may also appear at the Fairness Hearing either in person or through counsel retained at the objector's expense. Objectors or their attorneys intending to appear at the Fairness Hearing must follow procedures in the Class Notice Any objector who does not timely file and serve a notice of intention to appear in accordance with this paragraph and the procedures set forth in the Class Notice shall not be permitted to appear at the Fairness Hearing, except for good cause shown.
- 9. <u>Response to Objectors</u>. Class Counsel shall respond to any timely-filed and served objection by seven (7) calendar days before the Fairness Hearing.
- 10. <u>Notice Expenses</u>. The expenses of publication, printing and mailing of all notices required hereby, and the toll-free number charges described on Exhibit 1 hereto, shall be paid from the Settlement Fund as provided in Section 8.1 of the Settlement Agreement. Such expenses shall not be refundable to Defendants.

11. <u>Service of Papers</u>. Defendants' counsel and Class Counsel shall promptly furnish each other with copies of any and all objections that come into their possession.

12. <u>Additional Filings.</u> Class Counsel shall file no later than twenty-one (21) days prior to the Fairness Hearing motions for final approval of the Settlement, approval of the proposed Plan of Allocation, and applications for awards of attorneys' fees and cost reimbursements and Named Plaintiff compensation awards.

13. <u>Termination of Settlement</u>. This Order shall become null and void, and shall be without prejudice to the rights of the parties, all of whom shall be restored to their respective positions existing on December 15, 2008, if the Settlement Agreement is terminated in accordance with Section 10 thereof. In such event, Section 10 of the Settlement Agreement shall govern the rights of the parties.

14. <u>Use of Order</u>. In the event this Order becomes of no force or effect, it shall not be construed or used as an admission, concession, or declaration by or against the Defendants, the Named Plaintiffs or the Settlement Class.

15. <u>Continuance of Hearing</u>. The Court may continue the Fairness Hearing without further written notice.

SO ORDERED this

*** *** *** ***
Hon. Alvin W. Thompson, U.S.D.J.
11011. 7 11 viii vv. 1 110111p3011, O.D.D.J.

day of

, 2009.

EXHIBIT 1

TO

ORDER PRELIMINARILY APPROVING SETTLEMENT AND CONFIRMING FINAL SETTLEMENT HEARING

(SCOPE OF WORK DESCRIPTION FOR GILARDI & CO LLC)



January 26, 2009

Beth Leland, Esq. Keller Rohrback LLP 1201 Third Avenue, Suite 3200 Seattle, Washington 98101

Matter: In re Xerox Corporation ERISA Litigation

Gilardi & Co. LLC Statement of Work

BACKGROUND

As background, Gilardi is the nation's largest, private full-service administrator of class action settlements. For nearly a quarter-century, Gilardi has been the industry leader in class action settlement administration, with more than 3,000 cases administered and \$12 billion in assets distributed. On an annual basis, we provide escrow agent services for over 900 Qualified Settlement Funds and manage assets in excess of \$1.5 billion dollars. We are experts in all phases of legal administration and pride ourselves on our unmatched service, efficiency and accountability.

Additionally, Gilardi has been consistently ranked #1 or #2 by the RiskMetrics Group as the "Most Frequent Claims Administrators in the Top 100 Settlements" in securities litigation ad has been a national contract holder with the FTC for nearly a quarter century, during which time we have administered all aspects of the FTC's redress and distribution process.

SCOPE OF SERVICES

This Statement of Work documents the scope of service and deliverables in relation to the administration for the Xerox ERISA Litigation.

Gilardi will be engaged to perform Administration Services for the *In re Xerox Corporation ERISA Litigation*, in accordance with the terms of the class action Settlement Agreement, the Plan of Allocation and all orders of the Court.

Gilardi services will be performed as further described in Exhibit A, attached, to the extent applicable to the Scope of Services, described herein.

The steps of the Administration Services are outlined on the following page:

In re Xerox Corporation ERISA Litigation 2

Notification & Professional Services

- Format and draft Notice
- Format and publish Summary Notice
- Case coordination
- Apply National Change of Address update to class list
- Returned Undeliverable Mail (RUM) Handling
- Review 1997 Documentation

Administration

- Program case software based on Plan of Allocation calculation matrix
- Process and load data set into case software
- Perform data integrity check on calculation data set
- Set up Toll-Free Number, Message Script Recording with Rollover to Live Operators and After-Hours Voice Mail Box

Distribution

- Calculate pro-rata distribution amount to participants based on available distribution amount
- Audit distribution file and resolve any data or calculation issues

Confidentiality Statement

Gilardi and Co. LLC ("Gilardi") acknowledges that during the engagement Gilardi will have access to and become acquainted with confidential information, in both oral and written format (hereinafter "Confidential Information"), including but not limited to records and specifications owned, licensed or merely in the possession of Xerox Corporation (the "Company"); its sponsored employee benefit plans (including without limitation individually identifiable plan participant data (such as names, addresses, social security numbers, employment data, and participant transactions); and its or its plans' agents (the Company's plans and the agents of the Company and/or its plans collectively referred to as "Company-Related Parties"). Such information may include, without limitation the Company's and/or Company-Related Parties' processes, methods, customer lists, accounts and procedures. Gilardi agrees that we will not disclose any of the aforesaid, directly or indirectly, or use any of them in any manner, either during the term of this Agreement or at any time thereafter, except as required in the course of this engagement. All files, records, documents, specifications, information, letters, notes, media lists, account information and similar items relating to the business of the Company or any Company-Related Party, whether prepared by Gilardi or otherwise coming into Gilardi's possession, the Company or any Company-Related Party shall remain confidential and will only be disclosed as necessary the course of this engagement. Gilardi further agrees that we will at all times preserve the confidential nature of our relationship to the Company.

The confidentiality obligations in this agreement will not apply to the extent that Gilardi can demonstrate that:

- (a) The Confidential Information of the Company or any Company-Related Party at the time of disclosure was a part of the public domain;
- (b) The Confidential Information of the Company or any Company-Related Party became part of the public domain, by publication or otherwise, except by breach of the provisions of this Agreement;

Case 3:02-cv-01138-AWT Document 340-2 Filed 01/26/2009 Page 12 of 25

In re Xerox Corporation ERISA Litigation 3

- (c) The Confidential Information of the Company or any Company-Related Party can be established by written evidence to have been in the possession of the Gilardi at the time of disclosure;
- (d) The Confidential Information of the Company or any Company-Related Party is received by Gilardi from a third party without similar restrictions and without breach of this Agreement;
- (e) The Confidential Information of the Company or any Company-Related Party was developed by Gilardi or its agents independently of and without reference to any Confidential Information of the Company or Company-Related Parties.

We pride ourselves on being a trusted resource and on providing outstanding service to our clients. Please feel free to call Jenny Trang at (415) 458-3038 with any questions or concerns. We look forward to working with you on this project.

In re Xerox Corporation ERISA Litigation 4

EXHIBIT A – GILARDI'S PRIMARY ADMINISTRATION PHASES

Gilardi's primary service is the efficient execution of the four components of class action Due Process:

Notification • Processing • Distribution • Reporting

Our execution of these four steps is accomplished through an interactive team approach that involves our experienced staff and breadth of industry knowledge, Optical Character Recognition ("OCR") technology, proprietary software, and a handful of competitively priced, special purpose professional partners. You will find information detailing each of these primary phases below:

I. Case Set-up & Notification

Over the last 24 years, Gilardi has created a seamless, sophisticated process of identifying and notifying potential claimants nationwide. Because of our established, long-standing relationships with numerous brokers and nominees, we are able to facilitate the process of collecting potential claimant names and addresses, and to mail Notices and Proof of Claim Form packages in a timely manner. In so doing, we ensure that there is a proper and constant focus on the Due Process timing element of Notice. The Case Set-up & Notification phase includes the following services:

- ➤ Liaise with class counsel and Damage Expert to understand and/or assist in the design and development of the Settlement Fund Notice, Instructions and Claim Form
- Publish the Summary Notice
- > Solicit brokers/nominees to obtain potential claimant names and addresses
- > Print and mail Notice. Instructions and Claim Form
- Create a toll-free number to manage claimant inquiries
- > Track and maintain records of all opt outs
- Program case software to Plan of Allocation ("POA")
- > Design, implement and maintain a case website with links to the Notice, Claim Form and Frequently Asked Questions

II. Claims Processing

Gilardi is cost efficient and timely in processing claims. Our experienced staff has an unmatched breadth of industry knowledge to help class counsel understand and account for all the anomalies and nuances of case administration. Our processing procedures require that all cases undergo a variety of internal checks and balances to ensure a quality and accurate administration is completed. In addition, internal audit procedures are continuously refined to ensure we are actively managing the risks associated with processing a claim. Key components of this phase include the following:

- Open and prepare claims for scanning
- > Scan claims and supporting documentation via OCR technology
- Verify and proof claims
- ➤ Calculate all POA scenarios for accuracy and completeness
- Review and validate large claims, potential duplicate claims and security errors
- Research corporate actions, stock splits and merger/acquisition history

In re Xerox Corporation ERISA Litigation 5

EXHIBIT A – GILARDI'S PRIMARY ADMINISTRATION PHASES (CONT.)

- > Conduct independent audit
- Notify and resolve deficient claims and possible reject claims
- Notify class counsel when claims processing is complete
- > Check case database against known fraud database

III. Distribution

Gilardi provides a robust Fund Accounting Service for each individual administration. Distribution phase procedures undergo a variety of internal checks and balances to ensure accuracy and completeness. Multiple internal audit checks are performed at all phases of disbursement—check printing, folding, preparation, and mailing—to verify the data on the checks against the check registry and primary case software. Our daily communication with banks ensures an unsurpassed level of security and provides the foundation of our fraud detection and correction process.

Distribution services include:

- > Determine the distribution ratio
- > Open fund accounts
- ➤ Invest monies in short-term securities
- Prepare funds analysis
- > Prepare quarterly tax payments for Settlement Fund
- Calculate each claimant's pro-rata share of the Fair Fund
- > Print and mail distribution checks
- Follow-up on outstanding checks and reissue, as deemed necessary

IV. Reporting

Throughout the lifecycle of the administration, Gilardi will provide periodic updates to class counsel regarding the status of the fund administration. Gilardi will also prepare and submit all quarterly tax compliance reports of the settlement fund. Upon completion of the administration, we will create an affidavit summarizing our efforts, identifying the specifics of the case as well as provide a series of exhibits of the underlying data. A brief outline of some of the reportable statistics, include the following:

- Media plan execution with affidavits of publication and tear sheets
- ➤ Undeliverable and returned mail statistics
- Claim and case processing methodology
- Claim-specific loss calculation analyses
- > Claim counts and other aggregate case statistics
- > Settlement fund analyses
- > Distribution formulas, amounts and check counts
- ➤ Bank reconciliation detail, check reissues and tax filings

EXHIBIT 2

TO

ORDER PRELIMINARILY APPROVING SETTLEMENT AND **CONFIRMING FINAL SETTLEMENT HEARING**

(NOTICE OF PROPOSED SETTLEMENT OF ERISA CLASS ACTION LITIGATION, SETTLEMENT FAIRNESS HEARING, AND MOTION FOR: ATTORNEYS' FEES AND REIMBURSEMENT OF **EXPENSES AND CASE CONTRIBUTION AWARDS)**

Case 3:02-cv-01138-AWT Document 340-2 Filed 01/26/2009 Page 16 of 25 UNITED STATES DISTRICT COURT DISTRICT OF CONNECTICUT

In re XEROX CORPORATION ERISA LITIGATION

Master File No. 02-CV-1138 (AWT)

CLASS ACTION

CLASS ACTION

NOTICE OF PROPOSED SETTLEMENT OF ERISA CLASS ACTION LITIGATION, SETTLEMENT FAIRNESS HEARING, AND MOTION FOR: ATTORNEYS' FEES AND REIMBURSEMENT OF EXPENSES AND CASE CONTRIBUTION AWARDS

Your legal rights might be affected if you are a member of the following class:

All Actions

All current and former participants and beneficiaries of The Xerox Corporation Savings Plan or The Savings Plan of Xerox Corporation and the Xerographic Division, UNITE HERE (Formerly "The Savings Plan of Xerox Corporation and the Xerographic Division, Union of Needletrades, Industrial and Textile Employees, A.F.L.-C.I.O.-C.L.C.") (collectively, the "Plans") for whose individual accounts the Plans purchased and/or held interests in the Xerox Stock Fund at any time during the period May 12, 1997 through and including June 28, 2002, excluding the Individual defendants in the above-captioned litigation (collectively, the "Settlement Class," or the "Class").

PLEASE READ THIS NOTICE CAREFULLY. A FEDERAL COURT AUTHORIZED THIS NOTICE. THIS IS NOT A SOLICITATION. YOU HAVE NOT BEEN SUED.

This notice ("Notice") advises you of a proposed settlement (the "Settlement") of a consolidated class action lawsuit (the "Action") concerning the Plans' holding of and investment in Xerox common stock. The Action alleges, among other things, that the Defendants violated the Employee Retirement Income Security Act of 1974 ("ERISA") in connection with the Plans' holdings of and investment in Xerox stock in the Xerox Stock Fund option of the Plans. This Action concerns only claims for alleged violations of ERISA with respect to the Plans' holdings of and investment in Xerox stock; the Action does not concern claims or proceedings brought under the federal securities laws, state securities laws, or by the Securities and Exchange Commission. This Notice contains summary information about the Settlement. The full terms and conditions of the Settlement are set forth in a Class Action Settlement Agreement (the "Settlement Agreement"), which, along with the current Complaint and certain other materials pertaining to the Action and the Settlement, are available at www.KellerSettlements.com/XeroxERISA.html. In addition, Co-Lead Counsel for the Class, Keller Rohrback L.L.P. and Futterman Howard Watkins Wylie & Ashley, Chtd., have established a toll-free phone number, _______, and an email address, XEROXERISASettlement@kellerrohrback.com, if you have questions or need additional information. Please do not contact the Court, Xerox Corporation ("Xerox") or the Xerox Benefits Center with questions.

The Settlement provides for a cash payment by Defendants of \$51,000,000.00 and for allocation of this payment, net of expenses described in the Settlement Agreement (which include notice and allocation expenses, Court-approved attorneys' fees and expenses and Plaintiff compensation awards, taxes and other costs related to the Settlement Fund administration) to the Plan accounts of members of the Settlement Class who suffered losses as the result of investment of their Plan accounts in the Xerox Stock Fund. In addition, the Settlement contains provisions relating to the ongoing administration of the Plans and management of their assets.

The Court in charge of this case has scheduled a Fairness Hearing for __:___.m. on ________, 2009, to consider whether to grant final approval of the Settlement, and to address Co-Lead Counsel's motion for awards of attorneys' fees and expenses and their request for Plaintiff compensation awards. The Fairness Hearing will be held before the Honorable Alvin W. Thompson in the United States District Court for the District of Connecticut, Abraham Ribicoff Federal Building, 450 Main Street, Hartford, CT 06103.

Reasons for the Settlement: The reasons for the Settlement are the Parties' desires to resolve all claims in the Action against the Defendants for alleged violations of ERISA and end the Action by providing for payments to affected Settlement Class members. The Settlement is not, and should not be construed as, an admission of any fault, liability or wrongdoing whatsoever by any of the Defendants, who deny all fault, liability and wrongdoing. The Named Plaintiffs and Co-Lead Counsel believe that the proposed Settlement is fair, reasonable, and adequate, in the best interests of the Settlement Class, and preferable to continued litigation. The Plaintiffs and Co-Lead Counsel believe that it provides substantial benefits that would not be available in its absence.

PLEASE READ THIS NOTICE CAREFULLY AND COMPLETELY. IF YOU ARE A MEMBER OF THE SETTLEMENT CLASS TO WHOM THIS NOTICE IS ADDRESSED, THE SETTLEMENT WILL AFFECT YOUR RIGHTS. YOU ARE NOT BEING SUED IN THIS MATTER. YOU DO NOT HAVE TO APPEAR IN COURT, AND YOU DO NOT HAVE TO HIRE AN ATTORNEY IN THIS CASE. IF YOU ARE IN FAVOR OF THE SETTLEMENT, YOU NEED NOT DO ANYTHING. IF YOU DISAPPROVE, YOU MAY OBJECT TO THE SETTLEMENT PURSUANT TO THE PROCEDURES DESCRIBED BELOW.

ACTIONS YOU MAY TAKE IN THE SETTLEMENT

Case 3:02-cv-01138-AWT Documen	t 340-2 Filed 01/26/2009 Page 17 of 25
NO ACTION IS NECESSARY TO RECEIVE PAYMENT.	If the Settlement is approved by the Court and you are a member of the Settlement Class, you do not need to do anything in order to receive a payment. The portion, if any, of the Settlement Fund to be allocated to your Plan account will be calculated as part of the implementation of the Settlement. However, if you wish that transactions involving the Xerox Stock Fund in your Plan account that occurred during the period May 12, 1997 through December 31, 1997, be taken into account in calculating your share, if any, of the Settlement Fund, you must provide required documentation to the Settlement Administrator; please review carefully the discussion on Page of this Notice entitled "Special Information Regarding 1997 Plan Activity."
YOU CAN OBJECT NO LATER THAN, 2009.	If you wish to object to any part of the Settlement, you can follow the procedures stated below, and explain why you do not like the Settlement.
YOU CAN GO TO THE FAIRNESS HEARING ON, 2009 BY FILING A NOTICE OF INTENTION TO APPEAR NO LATER THAN, 2009.	You can ask to speak in Court about the fairness of the Settlement at the Fairness Hearing.

WHAT THIS NOTICE CONTAINS

SUMMARY OF SETTLEMENT
BASIC INFORMATION
1. Why did I get this Notice package?
2. How do I know whether I am part of the Settlement?
3. What does the Settlement provide?
4. What will be my share of the Settlement Fund?
5. What is the lawsuit about? What has happened so far?
6. Why is this case a class action?
7. Why is there a Settlement?
8. How can I get my portion of the recovery?
9. When will I receive my payment?
10. What rights am I giving up in the Settlement?
11. Can I exclude myself from the Settlement?
THE LAWYERS REPRESENTING YOU.
12. Do I have a lawyer in the case?
13. How will the lawyers be paid?
OBJECTING TO THE SETTLEMENT OR THE ATTORNEYS' FEES
14. How do I tell the Court if I don't like the Settlement?
THE COURT'S FAIRNESS HEARING
15. When and where will the Court decide whether to approve the Settlement?
16. Do I have to come to the hearing?
17. May I speak at the hearing?
IF YOU DO NOTHING
18. What happens if I do nothing at all?
GETTING MORE INFORMATION
19 How do I get more information?

Case 3:02-cv-01138-AWT Document 340-2 Filed 01/26/2009 Page 18 of 25

SUMMARY OF SETTLEMENT

Under the Settlement, Defendants will pay \$51,000,000 into a Settlement Fund and the Action will be dismissed. The Settlement Fund, including any accrued interest but net of expenses (including attorneys fees and expenses as awarded by the Court and other amounts as described above) will be allocated among Settlement Class members in accordance with a Plan of Allocation as approved by the Court. (See Section 4 below for details of the Plan of Allocation.) Co-Lead Counsel for the Class recommend and support the Settlement as a fair, reasonable and adequate compromise of the Action.

As with any litigation, the Parties would face an uncertain outcome if the Action were to continue against the Defendants. The Court has not ruled on the validity of the claims asserted in the Action. Continued litigation of the Action could result in a judgment greater or less than the recovery under the Settlement Agreement, or in no recovery at all. This litigation has been hotly contested from the outset. Indeed, throughout this litigation, the Named Plaintiffs and the Defendants have disagreed on both liability and damages, and they do not agree on the amount that would be recoverable even if the Plaintiffs were to prevail at trial. The Defendants, among other things: (1) have denied, and continue to deny, the material allegations of the Complaint; (2) have denied, and continue to deny, any wrongdoing or liability whatsoever; (3) believe that they acted at all times reasonably and prudently with respect to the Plans, their participants and beneficiaries, and the Settlement Class; (4) would assert certain other defenses if this Settlement is not consummated; and (5) are entering into the Settlement solely to avoid the cost, disruption, and uncertainty of litigation. Nevertheless, the Parties have taken into account the uncertainty and risks inherent in this litigation, particularly its complex nature, and have concluded that it is desirable that the Action be fully and finally settled on the terms and conditions set forth in the Settlement Agreement.

Co-Lead Counsel will apply to the Court for an order awarding attorneys' fees not in excess of 30% of the gross amount recovered in the Settlement, plus reimbursement of expenses. The Named Plaintiffs in the Action will share in the allocation of the money paid to the Plans on the same basis and to the same extent as all other members of the Settlement Class, except that, in addition, the Named Plaintiffs will apply to the Court for additional compensation of up to \$5,000 each for their efforts in prosecuting the Action since 2002. Any compensation awarded to the Named Plaintiffs by the Court will be payable from the proceeds of the Settlement.

BASIC INFORMATION

1. Why did I get this Notice package?

Either you or someone in your family appears to have been a participant in or beneficiary of one of the Plans and to have had a portion of your, his, or her Plan account(s) invested in the Xerox Stock Fund during the Class Period. The Court has directed that this Notice be sent to you because, as a potential Settlement Class member, you have a right to know about the proposed Settlement before the Court decides whether to approve it. If the Court approves the Settlement, and all objections and appeals are favorably resolved, the net amount of the Settlement Fund will be allocated among Class members according to a court-approved Plan of Allocation.

This Notice explains the Action, the Settlement, your legal rights, what benefits are available, who is eligible for them, and how you will receive your portion of the benefits. In addition, this Notice informs you of a hearing (the "Fairness Hearing") to be held at __:____m. on ______, 2009 before the Honorable Alvin W. Thompson in the United States District Court for the District of Connecticut, Abraham Ribicoff Federal Building, 450 Main Street, Hartford, CT 06103 to consider the fairness, reasonableness and adequacy of the proposed Settlement and to consider the application of Co-Lead Counsel for their attorneys' fees and reimbursement of litigation expenses as well as an application for case contribution awards to the Named Plaintiffs.

The issuance of this Notice is not an expression of the Court's opinion on the merits of any claim in the Action. If the Court approves the Settlement, payment to the Class will be made after all objections and appeals, if any, are favorably resolved. If there are any objections or appeals, resolving them can take time, so please be patient.

Case 3:02-cv-01138-AWT Document 340-2 Filed 01/26/2009 Page 19 of 25 2. How do I know whether I am part of the Settlement?

You are a member of the Settlement Class if: (a) you were a participant in or beneficiary of The Xerox Corporation Savings Plan or The Savings Plan of Xerox Corporation and the Xerographic Division, UNITE HERE (Formerly "The Savings Plan of Xerox Corporation and the Xerographic Division, Union of Needletrades, Industrial and Textile Employees, A.F.L.-C.I.O.-C.L.C.") (collectively, the "Plans"), (b) your individual accounts in the Plans held interests in the Xerox Stock Fund at any time during the period May 12, 1997 through and including June 28, 2002, and (c) you are not a Defendant in the Action.

3. What does the Settlement provide?

A Settlement Fund consisting of \$51,000,000.00 in cash, plus interest, is being established in connection with the Settlement of the Action, and for dismissal of the Action. If the Settlement is approved as described above, the net amount in the Settlement Fund (after payment of, and establishment of reserves for, certain amounts as described in the Settlement Agreement, including expenses associated with Class Notice, attorneys' fees and expenses, case contribution awards, taxes and other costs related to the administration of the Settlement Fund) will be allocated among and paid to members of the Class according to a Plan of Allocation to be approved by the Court. Allocations will be made to accounts established for members of the Class. Disbursement of the Settlement Fund to the Class will occur once the Settlement has become Final – after all appeals relating to the Settlement are favorably decided and all appeal periods have run. In addition, the Settlement contains provisions relating to the ongoing administration of the Plans and management of their assets.

This Action and the Settlement of it concern only alleged violations of ERISA with respect to Xerox stock held in the two ERISA Plans described herein. This Action does not involve Xerox stock purchased or held outside these Plans, or any other retirement or benefit plan, and does not involve any Xerox bonds.

The above description of the Settlement is only a summary. The full governing provisions are set forth in the Settlement Agreement (including exhibits), which may be obtained at www.KellerSettlements.com/XeroxERISA.html, or upon request of Co-Lead Counsel.

4. What will be my share of the Settlement Fund?

Your share of the Settlement Fund, if any, will be determined in accordance with a Plan of Allocation approved by the Court. The proposed Plan of Allocation, to be considered by the Court at the Fairness Hearing, is attached to the Settlement Agreement as Exhibit C. The proposed Plan of Allocation sets forth a methodology that takes into account the value of any investments in Xerox stock in your account(s) at the beginning and end of the "Allocation Period," generally January 2, 1998 through June 28, 2002 (the beginning of the Allocation Period may be advanced to as early as May 12, 1997, in the cases of persons who provide adequate information with respect to 1997 acquisitions of Xerox stock in the Plan accounts, see "Special Information Regarding 1997 Plan Activity" below), as well as the value of investments in the Xerox Stock Fund in your account(s) (whether as a result of Company matching or voluntary contributions), and the proceeds of sales or withdrawals from the Xerox Stock Fund in your account(s) during the Allocation Period.

The calculations necessary to determine your share of the Net Proceeds, if any, will be made for you in connection with the administration of the Settlement and will determine both whether you are entitled to a share of the Net Proceeds and your share amount. Those performing the calculations will have access to all relevant and available Plan records and so, subject to the discussion immediately below with respect to 1997 Plan activity, you do not need to be concerned if you no longer have your Plan account statements, and do not need to supply anyone with statements or records. The Court will be asked to approve a more detailed statement of the Plan of Allocation, a copy of which is available along with other settlement documents at www.KellerSettlements.com/XeroxERISA.html.

SPECIAL INFORMATION REGARDING 1997 PLAN ACTIVITY: If you have account records generated by Xerox or other contemporaneous written documentation showing holdings, purchases or sales of the Xerox Stock Fund in your Plan account during the period from May 12, 1997 through December 31, 1997, you may, but are not required to, submit such materials to the Settlement Administrator on or before -------, 2009, by mailing them to the Settlement Administrator at *In re Xerox Corporation ERISA Litigation*, c/o Gilardi & Co. LLC, P.O. Box 8040, San Rafael, CA 94912-8040, first class postage prepaid. If you submit such records, please retain a copy for your files. Such records will, to the extent they are deemed reliable by the Settlement Administrator, be considered in calculating your share of the Settlement Fund. The submission of such records may or may not affect the amount you receive from the Settlement; if they do affect it, they may increase, or reduce, your share of the Settlement Fund. You are under no obligation to submit 1997 materials if you do not wish for your 1997 Plan activity to be taken into account. Please do not contact the Plans, Xerox, or the Xerox Benefits Center requesting these records because they do not have them and will not be able to provide them. Note, handwritten notes and similar personal records are not acceptable. The only acceptable records are records generated by Xerox or other contemporaneous written documentation deemed reliable by the Settlement Administrator.

5. What is the lawsuit about? What has happened so far?

The Action is a consolidation of several putative class action lawsuits that were brought on behalf of a purported class of individuals who were participants in or beneficiaries of the Plans during the Class Period, alleging violations of ERISA in connection with the Plans' and their participants' holdings in the Xerox Stock Fund option of the Plans. The Action has been pending since 2002.

Plaintiffs allege among other things, that the Defendants were fiduciaries of the Plans and violated their obligations under ERISA by causing the Plans to imprudently offer, purchase and hold units of the Xerox Stock Fund when Xerox stock was allegedly an unsuitable investment for the Plans. Plaintiffs further allege that the Defendants misrepresented to Plan participants the true financial condition of the Company and, consequently, the true value of the Xerox Stock Fund. Plaintiffs sought to recover from the Defendants losses to the Plans and their participants such as yourself caused by the Defendants' alleged misconduct.

Case 3:02-cv-01138-AWT Document 340-2 Filed 01/26/2009 Page 20 of 25

The Defendants have vigorously denied and continue to deny all of the claims and allegations against them. Defendants maintain that their actions and statements were in all respects lawful, prudent, proper and truthful, and deny that anyone has been harmed in any way by any action or inaction on their part.

Co-Lead Counsel have conducted extensive legal research and discovery regarding the facts and claims in this Action, including reviewing millions of pages of documents produced in this Action by Xerox and other entities, including but not limited to Xerox's current and former auditors, and the Plans' trustee, recordkeeper, and investment manager; conducting numerous depositions of past and present Xerox officers, directors and employees; and analyzing studies and reports authored by expert consultants retained on the Class' behalf regarding a variety of matters, including estimated potential damages in the Action.

The Parties have also engaged in substantial motion practice, i.e., briefing and legal argument, before the Court. The Court has entered two opinions on Defendants' motions to dismiss the Action, granting these motions in part, and denying them in part. In addition, a motion to certify the Action as a class action has been fully briefed and was pending undecided when the Settlement was reached.

After engaging in these litigation efforts, and negotiating for several months using an independent paid mediator, Co-Lead Counsel for Plaintiffs and counsel for the Defendants reached an agreement to settle the Action, subject to Court approval, on the terms summarized in this Notice.

6. Why is this case a class action?

It is a class action because the claims of similarly situated persons can be resolved more efficiently in one proceeding than in many and because the claims of individual Plan participants are generally too small to warrant litigation unless they can be aggregated. In a class action, one or more plaintiffs, called "named plaintiffs," sue on behalf of people who have similar claims. All of the individuals on whose behalf the Named Plaintiffs in this Action are suing are "Class members," also referred to in this Notice as members of the Settlement Class.

7. Why is there a Settlement?

So that the Class can be assured of a prompt, fair and substantial recovery and avoid the expenses, risks and delays of further contested litigation. Under the proposed Settlement, the Court will not decide the merits of the Action in favor of either the Plaintiffs or the Defendants; instead they will be compromised and resolved without a trial.

Co-Lead Counsel believe that the proposed Settlement is fair, reasonable and adequate and in the best interest of the Class.

8. How can I get my portion of the recovery?

Subject to the discussion on Page ___ under "Special Information Regarding 1997 Plan Activity," Members of the Class do not need to do anything to receive their share of the recovery in this Action.

9. When will I receive my payment?

It will be six months at the very least, and maybe considerably longer. Payment is conditioned on several matters, including the Court's approval of the Settlement and that approval becoming Final and no longer subject to any appeals. If there are objections or appeals, the process could take one or more years. Upon satisfaction of the applicable conditions, and assuming the Settlement is not terminated according to its terms, the Net Proceeds will be allocated in accordance with the Plan of Allocation and credited to Class Members' Plan accounts. Class members who no longer have an active account with the Plans will have an account created for purposes of receipt of Settlement funds. Class members will receive communications regarding their accounts as described herein and will be entitled to distributions of their account balances in accordance with the terms of the Plans. Any accrued interest on the Settlement Fund will be included in the amount allocated and paid to the Class members.

10. What rights am I giving up in the Settlement?

You are giving up your right to sue the Defendants for anything in any way related to the Plans' investment in Xerox stock during the Class Period. If the Settlement is approved, the Court will enter a judgment that will end and dispose of the Action and all the claims asserted in it. It will also fully, finally, and forever release, relinquish, and discharge all of the Released Parties (generally, the Defendants and certain persons affiliated with the Defendants or fiduciaries of the Plans, all as set forth in the Settlement Agreement) from any and all Released Claims (also as defined in the Settlement Agreement, generally the claims that were or could have been asserted in the Action or would have been barred by law had the Action proceeded to final judgment in favor of the Defendants).

11. Can I exclude myself from the Settlement?

No. You do not have the right to exclude yourself from the Settlement. The Action was certified under Federal Rule of Civil Procedure 23(b)(1). Thus, it is not possible for any of the Class members to exclude themselves from the Settlement. As a member of the Class, you will be bound by any judgments or orders that are entered in the Action for all claims that were or could have been asserted in the Action against the Defendants or are otherwise included in the release under the Settlement.

THE LAWYERS REPRESENTING YOU

12. Do I have a lawyer in the case?

Yes. The Court has appointed the law firms of Keller Rohrback L.L.P. and Futterman Howard Watkins Wylie & Ashley, Chtd. as Co-Lead and Class Counsel for the Named Plaintiffs, the Plans, and the Class ("Co-Lead Counsel"). You will not be charged directly by these lawyers. If you want to be represented by your own lawyer, you may hire one, but it will be at your own cost and expense.

13. How will the lawyers be paid?

At the Fairness Hearing, Co-Lead Counsel will apply for an award of attorneys' fees and expenses on behalf of all Plaintiffs' counsel. The

Case 3:02-cv-01138-AWT Document 340-2 Filed 01/26/2009 Page 21 of 25

application for attorneys' fees will not exceed 30% of the gross amount recovered in the Settlement, plus reimbursement of expenses. Any award of fees and expenses will be paid from the Settlement Fund prior to allocation and payment to the Class members.

To date, neither Co-Lead Counsel nor any additional counsel have received any payment for their services in prosecuting this Action on behalf of the Class since the Action began in 2002, nor have counsel been reimbursed for any of their out-of-pocket expenses. To date, counsel working on behalf of the Class have expended in excess of -----hours of uncompensated professional time and over \$------ in unreimbursed expenses of litigation. The fee requested by Co-Lead Counsel would compensate all counsel who have worked on behalf of the Class for their efforts in achieving the Settlement and for the risk they have borne in undertaking this representation on a contingency basis. The Court will determine the amount of the fee and expense award.

OBJECTING TO THE SETTLEMENT OR THE ATTORNEYS' FEES

14. How do I tell the Court if I like or don't like the Settlement or the request for attorneys' fees and expenses?

If you are in favor of the Settlement, you do not need to do anything. If you oppose the Settlement, or the request for attorneys fees and expenses or the request for Named Plaintiff awards, you may object. To object, you must file with the Court a written objection that is labeled "Objection" and that states the name of this Action: *In re Xerox ERISA Litigation*, Master File No.: 02-CV-1138 (AWT). Your objection must state all the bases and reasons for the objection, set forth proof of your membership in the Settlement Class, clearly identify any and all witnesses, documents and other evidence of any kind that are to be presented at the Fairness Hearing or otherwise in connection with such objection, and further describe the substance of any testimony to be given either by you or any supporting witnesses. To file your objection you must send it to the Court or file it with the Court Be sure to include your name, address, telephone number and signature. **Your written Objection must be filed with the Court, and mailed to the counsel listed below, postmarked and sent via facsimile to counsel listed below on or before 5:00 p.m., Eastern Time, on , 2009:**

To file with the Clerk of the Court, mail or hand-deliver to:

Clerk of the Court United States District Court for the District of Connecticut Abraham Ribicoff Federal Building 450 Main Street, Hartford, CT 06103

To serve by mail and facsimile, mail and fax or hand-deliver to each of the following:

ERISA CO-LEAD COUNSEL FOR THE PLAINTIFFS:

Lynn Lincoln Sarko KELLER ROHRBACK L.L.P. 1201 Third Avenue, Suite 3200 Seattle, WA 98101 Fax: (206) 623-3384

WYLIE & ASHLEY, CHTD. 122 S. Michigan Ave., Suite 1850 Chicago, IL 60603 Fax: (312) 427-1850

FUTTERMAN HOWARD WATKINS

Charles R. Watkins

COUNSEL FOR DEFENDANTS:

Steven J. Sacher JONES DAY 51 Louisiana Ave. N.W. Washington, DC 20001-2113 Fax: (202) 626-1700

John A. Valentine WILMER CUTLER PICKERING HALE AND DORR LLP 1875 Pennsylvania Ave., NW Washington, DC 20006 Fax: (202) 663-6363 Michael T. Hannafan HANNAFAN & HANNAFAN, LTD. One East Wacker Drive, Suite 2800 Chicago, IL 60601 Fax: (312) 527-0220

Thomas D. Goldberg DAY PITNEY LLP One Canterbury Green Stamford, CT 06901 Fax: (203) 977-7301

UNLESS OTHERWISE ORDERED BY THE COURT, ANY CLASS MEMBER WHO DOES NOT OBJECT IN THE MANNER DESCRIBED HEREIN WILL BE DEEMED TO HAVE WAIVED ANY OBJECTION AND SHALL BE FOREVER FORECLOSED FROM MAKING ANY OBJECTION TO THE PROPOSED SETTLEMENT AND THE APPLICATION FOR ATTORNEYS' FEES AND EXPENSES AND CASE CONTRIBUTION AWARDS.

THE COURT'S FAIRNESS HEARING

15. When and where will the Court decide whether to approve the Settlement?

At the Fairness Hearing. The Court will hold the Fairness Hearing at __:___.m. on _______, 2009, at the United States District Court for the District of Connecticut, Abraham Ribicoff Federal Building, 450 Main Street, Hartford, CT 06103. At the Fairness Hearing the Court will consider whether to approve the Settlement, and may consider any request for attorneys fees and expenses and the request for Plaintiff compensation awards.

If there are objections, the Court will consider them, as well. At or after the Fairness Hearing, the Court will rule on these matters. We do not

Case 3:02-cv-01138-AWT Document 340-2 Filed 01/26/2009 Page 22 of 25

know how long these rulings will take.

16. Do I have to come to the Fairness Hearing?

No. Co-Lead Counsel will represent you and handle the Fairness Hearing, and answer questions Judge Thompson may have at that time. You are welcome to come at your own expense, however. If you send an objection, you do not have to come to Court to talk about it. As long as you mailed your written objection on time and otherwise complied with the requirements of Paragraph 14 herein, it will be before the Court when the Court considers whether to approve the Settlement as fair, reasonable and adequate. You may also have your own lawyer attend the Fairness Hearing at your expense, but such attendance is also not necessary.

17. May I speak at the Fairness Hearing?

The Fairness Hearing may be delayed by the Court without further notice. If you wish to attend the Fairness Hearing, you may wish to confirm its date and time in advance with Co-Lead Counsel.

IF YOU DO NOTHING

18. What happens if I do nothing?

If you are a Class member, you will participate in the Settlement as described above in this Notice, provided the Settlement is approved.

GETTING MORE INFORMATION

19. How do I get more information?

This Notice summarizes the proposed Settlement. F	ull details of the Settlement are set forth in the Settlement Agreement. You may review
and print a copy of the Settlement Agreement by v	risiting www.KellerSettlements.com/XeroxERISA.html and may also obtain a copy by
making a written request to Co-Lead Counsel liste	ed on Page Other papers relating to the Settlement, such as the motion seeking
preliminary approval of the Settlement Agreement ar	nd the Preliminary Approval Order, may be viewed and obtained in similar manner. You
may direct inquiries to a toll-free phone number (), or via email sent to <u>info@XEROXERISASettlement.com</u> .
DATED: JANUARY, 2009	By Order of the Court

EXHIBIT 3

TO

ORDER PRELIMINARILY APPROVING SETTLEMENT AND CONFIRMING FINAL SETTLEMENT HEARING

(SUMMARY NOTICE OF SETTLEMENT FOR PUBLICATION)

UNITED STATES DISTRICT COURT DISTRICT OF CONNECTICUT

In re XEROX CORPORATION ERISA LITIGATION	Master File No. 02-CV-1138 (AWT)
This Document Relates To:	CLASS ACTION
All Actions	

TO ALL MEMBERS OF THE FOLLOWING CLASS:

All current and former participants and beneficiaries of The Xerox Corporation Savings Plan or The Savings Plan of Xerox Corporation and the Xerographic Division, UNITE HERE (Formerly "The Savings Plan of Xerox Corporation and the Xerographic Division, Union of Needletrades, Industrial and Textile Employees, A.F.L.-C.I.O.-C.L.C.") (collectively, the "Plans") for whose individual accounts the Plans purchased and/or held interests in the Xerox Stock Fund at any time during the period May 12, 1997 through and including June 28, 2002, excluding the Individual defendants in the above-captioned litigation (collectively, the "Settlement Class").

PLEASE READ THIS NOTICE CAREFULLY. THIS IS A COURT-ORDERED LEGAL NOTICE. THIS IS NOT A SOLICITATION.

The United States District Court for the District of Connecticut authorized this Notice.

A proposed Settlement has been preliminarily approved by the federal court presiding over the above-captioned class action lawsuit, which alleges violations of the Employee Retirement Income Security Act of 1974 ("ERISA") in connection with the above-mentioned Plans' holdings of Xerox common stock in Xerox's 401(k) plans between May 12, 1997 and June 28, 2002 (the "Class Period"). The proposed Settlement provides for a payment of \$51 million, net of expenses described in the Settlement Agreement (which include notice and allocation expenses, Court-approved attorneys' fees and expenses and Plaintiff compensation awards, taxes and other costs related to the Settlement Fund administration) to the Plan accounts of members of the Settlement Class who suffered losses as the result of investment of their Plan accounts in the Xerox Stock Fund. In addition, the Settlement contains provisions relating to the ongoing administration of the Plans and management of their assets.

Under the Settlement, the Settlement Fund, including any accrued interest but net of expenses will be allocated among Settlement Class members who had portions of their Plan accounts invested in the Xerox Stock Fund option of the Plans in accordance with a Plan of

Allocation as approved by the Court.

The Defendants in the lawsuit continue to deny any liability or wrongdoing and do not admit to any wrongdoing in the settlement. The full provisions of the Settlement are set forth in a Settlement Agreement that is available at www.KellerSettlements.com/XeroxERISA.html. If you qualify, you may receive an allocation of a portion of the Settlement funds. You do not need to send in a claim or take any other action to participate in the Settlement.

THE COURT WILL HOLD A HEARING AT _:_ _.M. ON _____ __, 2009 TO DECIDE WHETHER TO APPROVE THE SETTLEMENT.

ADDITIONAL INFORMATION CONCERNING THE PROPOSED SETTLEMENT, INCLUDING THE NOTICE OF CLASS ACTION SETTLEMENT WHICH HAS BEEN MAILED TO CLASS MEMBERS (WHICH CONTAINS, AMONG OTHER THINGS, SPECIAL INFORMATION REGARDING THE SETTLEMENT FOR PERSONS WHO WERE PARTICIPANTS IN THE PLANS DURING 1997), THE CLASS ACTION SETTLEMENT AGREEMENT AND OTHER DOCUMENTATION, IS AVAILABLE AT WWW.KELLERSETTLEMENTS.COM/XEROXERISA.HTML. IN ADDITION, CO-LEAD COUNSEL FOR THE PLAINTIFFS HAVE ESTABLISHED A TOLL-FREE NUMBER, ()**AND EMAIL** ADDRESS, XEROXERISASETTLEMENT@KELLERROHRBACK.COM, TO ASSIST IN ANSWERING QUESTIONS REGARDING THE SETTLEMENT. YOU MAY ALSO **CONTACT CO-LEAD COUNSEL AT:**

> Lynn Lincoln Sarko KELLER ROHRBACK L.L.P. 1201 Third Avenue, Suite 3200 Seattle, WA 98101 Fax: (206) 623-3384

Charles R. Watkins **FUTTERMAN HOWARD WATKINS** WYLIE & ASHLEY, CHTD. 122 S. Michigan Ave. **Suite 1850** Chicago, IL 60603 Fax: (312) 427-1850

Please direct questions to Co-Lead Counsel, and not to the Court or Xerox.

2009 By Order of the Court DATED:

EXHIBIT B TO CLASS ACTION SETTLEMENT AGREEMENT

(ORDER AND FINAL JUDGMENT)

UNITED STATES DISTRICT COURT DISTRICT OF CONNECTICUT

1

In re XEROX CORPORATION ERISA LITIGATION	Master File No. 02-CV-1138 (AWT)
This Document Relates To:	CLASS ACTION
All Actions	, 2009

ORDER AND FINAL JUDGMENT

This matter came before the Court for a hearing on _______, 2009, pursuant to Fed. R. Civ. P. 23(e) and the Order of this Court dated ________, 2009 (the "Preliminary Approval Order"), on the application of the parties for approval of the Settlement set forth in the Settlement Agreement, executed on January ____, 2009, on behalf of the Parties. Notice having been given to the Settlement Class as required in the Preliminary Approval Order, and the Court having considered the Settlement Agreement, all papers filed and proceedings held herein, and good cause appearing therefore, IT IS HEREBY ORDERED, ADJUDGED AND DECREED as follows:

- 1. Capitalized terms not otherwise defined in this Order shall have the same meaning as ascribed to them in the Settlement Agreement, a true and correct copy of which is attached hereto as Exhibit 1.
- 2. This Court has jurisdiction over the subject matter of this Action and over all parties to the Action, including all members of the Settlement Class.
- 3. The Court determines that Plaintiffs are asserting alleged violations of the Employee Retirement Income Security Act of 1974, as amended, 29 U.S.C. §§ 1001, et seq.

("ERISA"), set forth in Plaintiffs' Third Consolidated Amended Complaint dated July 1, 2008, with respect to The Xerox Corporation Savings Plan and The Savings Plan of Xerox Corporation and the Xerographic Division, UNITE HERE (Formerly "The Savings Plan of Xerox Corporation and the Xerographic Division, Union of Needletrades, Industrial and Textile Employees, A.F.L.-C.I.O.-C.L.C."), together with their predecessors and successors, and any trust created under such plans (the "Plans").

- 4. The Court determines that the Settlement has been negotiated vigorously and at arm's length by the Parties, that the Settlement arises from a genuine controversy between the Parties and is not the result of collusion, nor was the Settlement procured by fraud or misrepresentation.
- 5. Pursuant to Fed. R. Civ. P. 23, the Court hereby approves and confirms the settlement embodied in the Settlement Agreement as being a fair, reasonable and adequate settlement and compromise of the Action, adopts the Settlement Agreement as its judgment, and orders that the Settlement Agreement shall be herewith effective, binding, and enforced according to its terms and conditions.
- 6. The Court finds that the requirements of the Federal Rules of Civil Procedure, the United States Constitution, the Rules of the Court and any other applicable law have been met as to the Settlement Class, in that:
 - a. The Settlement Class is ascertainable from records kept with respect to the Plans and from other objective criteria, and the members of the Settlement Class are so numerous that their joinder before the Court would be impracticable.

Page 4 of 38

- b. There are one or more questions of fact and/or law common to the Settlement Class.
- The claims of the Named Plaintiffs are typical of the claims of the c. Settlement Class.
- d. The Named Plaintiffs will fairly and adequately protect the interest of the Settlement Class in that (i) the interests of Named Plaintiffs and the nature of their alleged claims are consistent with those of the members of the Settlement Class, (ii) there are no conflicts between or among the Named Plaintiffs and the Settlement Class, and (iii) the Named Plaintiffs and the members of the Settlement Class are represented by qualified, reputable counsel who are experienced in preparing and prosecuting large, complicated ERISA class actions.
- The prosecution of separate actions by individual members of the e. Settlement Class would create a risk of (i) inconsistent or varying adjudications as to individual class members, that would establish incompatible standards of conduct for the parties opposing the claims asserted in the Action or (ii) adjudications as to individual class members that would, as a practical matter, be dispositive of the interests of the other members not parties to the adjudications, or substantially impair or impede those persons' ability to protect their interests.

- 7. Notice of the pendency of this action as a class action and of its proposed settlement has been provided and made in accordance with the Preliminary Approval Order and the Court finds as follows:
 - such notices and the method by which they were provided to the
 Settlement Class were appropriate and reasonable;
 - such notices included individual notice to all members of the Settlement
 Class who could be identified through reasonable efforts;
 - c. such notices provided valid, due and sufficient notice of these proceedings and of the matters set forth therein, including the settlement described in the Settlement Agreement, and including information regarding the procedure for the making of objections by all persons to whom such notices were directed; and
 - d. such notices fully satisfied the requirements of Fed. R. Civ. P. 23 and the requirements of due process.
- 8. The Court finds that the Plans' participation in the Settlement is on terms no less favorable than Plaintiffs and the Settlement Class and that the Plans do not have any additional Claims against any Released Party above and beyond the Released Claims (i) that are, were or could have been alleged, asserted, or set forth in the Complaint or (ii) that would be barred by principles of res judicata had the claims asserted in the Complaint been fully litigated and resulted in a Final judgment or order. The Court further finds that the Plans engaged an independent fiduciary to review the Settlement, and that such fiduciary has authorized the Plans to give the release provided for in the Settlement Agreement. Accordingly, the Court determines

that the negotiation and consummation of the Settlement by Named Plaintiffs on behalf of the Plans and the Settlement Class do not constitute "prohibited transactions" as defined by ERISA § 406(a) or (b), 29 U.S.C. § 1106(a) or (b). Further, the Court finds that, to the extent any of the transactions required by the Settlement constitute a transaction prohibited by ERISA § 406(a), 29 U.S.C. § 1106(a), such transactions satisfy the provisions of Prohibited Transaction Class Exemption 2003-39, 68 Fed. Reg. 75632 (2003).

9. The Court hereby certifies the Settlement Class for settlement purposes as a nonopt-out class action pursuant to Federal Rules of Civil Procedure 23(a) and 23(b)(1), with the class being defined as follows:

all current and former participants and beneficiaries of the Plans for whose individual accounts the Plans purchased and/or held interests in the Xerox Stock Fund at any time during the period May 12, 1997 through and including June 28, 2002, excluding the Individual Defendants.

The Court finds that the Settlement Class is sufficiently well-defined and cohesive, and meets the applicable requirements for a settlement class imposed by Federal Rule of Civil Procedure 23.

- 10. Plaintiffs' Counsel appointed by this Court as conditional Class Counsel in the Preliminary Approval Order, are hereby appointed class counsel pursuant to Fed R. Civ. P. 23 (g).
- 11. The Action is hereby dismissed as against all Defendants, with prejudice, each party to bear his, her or its own costs, except as provided herein, in the Settlement Agreement, or in a separate order entered by the Court.
- 12. As of the effective date of this judgment, each member of the Settlement Class, on their own behalf and on behalf of those who directly, indirectly, derivatively, or in any other capacity ever had, now have or hereafter may have Released Claims, and the Plans and their

current and former fiduciaries, shall be deemed to have and shall have absolutely and unconditionally released and forever discharged with prejudice the Released Parties from all Released Claims.

- 13. All members of the Settlement Class and the Plans are permanently barred and enjoined from the institution and prosecution, either directly or indirectly, of any other actions in any court asserting any and all Released Claims against any Released Party.
- 14. By operation of this judgment, each of the Defendants shall be deemed to have, and shall have, absolutely and unconditionally released and forever discharged the Plaintiff Releasees from any and all Claims relating to the institution or prosecution of the Action or the settlement of any Released Claims, except that the release described in this paragraph shall not include claims relating to the covenants or obligations set forth in this Settlement Agreement or in the Amended Consent Protective Order entered by the Court in this Action on August 1, 2006.
- 15. The Plan of Allocation attached hereto as Exhibit __ is approved as fair and reasonable. The Court finds payments and distributions made in accordance with such Plan of Allocation to be "restorative payments" as defined in IRS Revenue Ruling 2002-45. Any modification or change in the Plan of Allocation that may hereafter be approved shall in no way disturb or affect this Judgment and shall be considered separate from this Judgment.
- 16. Neither the Settlement Agreement nor the terms of the Settlement Agreement shall be offered or received into any Action or proceeding for any purposes, except: (a) in an action or proceeding arising under this Settlement Agreement or arising out of or relating to the Preliminary Approval Order or the Final Order; (b) in any action or proceeding where the Releases provided pursuant to this Settlement Agreement may serve as a bar to recovery; or (c) in any action or proceeding to determine the availability, scope, or extent of insurance coverage

Case 3:02-cv-01138-AWT Document 340-3 Filed 01/26/2009 Page 8 of 38

(or reinsurance related to such coverage) for the sums expended for the settlement and defense of

this Action.

17. Without affecting the finality of this Judgment in any way, this Court hereby

retains continuing and exclusive jurisdiction over: (a) implementation of the Settlement and any

award or distribution of the Settlement Fund; (b) disposition of the Settlement Fund and

implementation of the Plan of Allocation; (c) hearing and determining applications for attorneys'

fees, costs, interest and reimbursement of expenses in the Action; and (d) all Parties hereto for

the purpose of construing, enforcing and administering the Settlement.

18. In the event that the Settlement Agreement is terminated in accordance with its

terms then this Judgment shall be rendered null and void to the extent provided by and in

accordance with the Settlement Agreement and shall be vacated, and in such event, all orders

entered and releases delivered in connection herewith shall be null and void to the extent

provided by and in accordance with the Settlement Agreement.

19. Final Judgment shall be entered herein.

SO ORDERED this day of 2009.

Hon. Alvin W. Thompson, U.S.D.J.

7

EXHIBIT 1 TO ORDER AND FINAL JUDGMENT

(CLASS ACTION SETTLEMENT AGREEMENT)

CLASS ACTION SETTLEMENT AGREEMENT

This CLASS ACTION SETTLEMENT AGREEMENT ("Settlement Agreement") is entered into by and between Named Plaintiffs (as defined below) in the action titled In re Xerox Corporation ERISA Litigation, Civ. No. 3:02-CV-01128 (AWT) (D. Conn.) for themselves and on behalf of the Settlement Class (as defined below) and the Plans (as defined below), on the one hand, and the *Defendants* (as defined below) on the other, in consideration of the promises, covenants and agreements herein described and for other good and valuable consideration acknowledged by each of them to be satisfactory and adequate.

NOW, THEREFORE, without any admission or concession on the part of the Named Plaintiffs of any lack of merit of the action, and without any admission or concession on the part of Defendants as to the merit of the action, it is hereby STIPULATED AND AGREED, by and among the Parties (as defined below) to this Settlement Agreement, through their respective attorneys, subject to approval of the Court pursuant to the Federal Rules of Civil Procedure, in consideration of the benefits flowing to the Parties hereto from the Settlement Agreement, that all Released Claims (as defined below) as against the Released Parties (as defined below) shall be compromised, settled, released and dismissed with prejudice, upon and subject to the following terms and conditions:

1. Definitions.

As used in this Settlement Agreement, italicized and capitalized terms and phrases not otherwise defined herein have the meanings provided below:

- "Action" shall mean: In re Xerox Corporation ERISA Litigation, Civ. No. 3:02-CV-01128 (AWT), United States District Court for the District of Connecticut, and any and all cases now or hereafter consolidated therewith.
- "Agreement Execution Date" shall mean: the date on which this Settlement Agreement is fully executed, as provided in Section 12.11 below.
- "Other Appointed Counsel" shall mean: McTigue & Porter LLP, Stember 1.3 Feinstein Doyle & Payne, LLC, Barroway Topaz Kessler Meltzer Check, LLP, Berger & Montague, P.C., Law Offices of Daniel M. Harris, and Goodman Rosenthal & McKenna PC. and their predecessors and Successors-In-Interest.
 - 1.4 "Claims" shall have the meaning set forth in Section 3.2.
- "Class Notice" shall mean: the forms of notice appended as Exhibits 2 and 3 to the form of Preliminary Approval Order, attached hereto as Exhibit A.
 - 1.6 "Class Period" shall mean: May 12, 1997 through and including June 28, 2002.

- 1.7 "Co-Lead Counsel" shall mean: Lynn Sarko, Keller Rohrback L.L.P. and Charles R. Watkins, Futterman Howard Watkins Wylie & Ashley, Chtd. (previously of Susman & Watkins and Susman, Watkins & Wylie).
- "Company" or "Xerox" shall mean: Xerox Corporation, each of Xerox's 1.8 predecessors and Successors-In-Interest; any Person that controls, is controlled by, or is under common control with *Xerox*; any of its direct and indirect parents and subsidiaries, and any company whose employees participated in either of the Plans. For the purposes of this Settlement Agreement only, one company "controls" another company if (i) the former company, directly or indirectly, or acting through one or more other persons owns, has owned, or has or had the power to vote more than 50 percent of any class of voting securities of the latter company; or (ii) the former company has or had the right to determine in any manner the election of a majority of the directors or trustees of the latter company.
- "Complaint" shall mean: the Third Consolidated Amended Complaint in the Action, filed July 1, 2008.
- "Court" shall mean: the United States District Court for the District of 1.10 Connecticut.
- "Defendants" shall mean the following persons and/or entities: Lawrence Becker, 1.11 Sally Conkright, Patricia M. Nazemetz, Arlyn B. Kaster, Myra R. Drucker, Kathleen Russell, William Strusz, Lance Davis, Gregory B. Tayler, Christina Clayton, Gary Kabureck, Lawrence Zimmerman, Paul A. Allaire, Anne M. Mulcahy, William F. Buehler, Barry D. Romeril, B.R. Inman, Vernon E. Jordan, Jr., Hilmar Kopper, George J. Mitchell, N.J. Nicholas, Jr., Patricia F. Russo, Martha R. Seger, Thomas C. Theobald, G. Richard Thoman, Barbara D. Roscoe as beneficiary of property passing pursuant to the Last Will and Testament of William C. Roscoe, dated December 30, 2005, Henry Charles Filter, III, as beneficiary of property passing pursuant to the Will of Eunice M. Filter dated February 25, 2002 to the Eunice M. Filter Revocable Trust dated February 25, 2002, Henry Charles Filter, III, and John Musicaro, Jr., as Trustees of the Trust for the benefit of Henry Charles Filter, III and his descendants under Article II of the Eunice M. Filter Insurance Trust under Agreement dated January 19, 1995, as beneficiary of property passing pursuant to the Will of Eunice M. Filter dated February 25, 2002 to the Eunice M. Filter Revocable Trust dated February 25, 2002, c/o John Musicaro, Jr., Cummings & Lockwood LLC, Six Landmark Square, Stamford, CT 06901, and Jerry W. Hostetter and John Musicaro, Jr. as Trustees of the Marital Trust under Article IV of the Eunice M. Filter Revocable Trust dated February 25, 2002, as beneficiary of property passing pursuant to the Will of Eunice M. Filter dated February 25, 2002 to the Eunice M. Filter Revocable Trust dated February 25, 2002, c/o John Musicaro, Jr., Cummings & Lockwood LLC, Six Landmark Square, Stamford, CT 06901 (collectively, the "Individual Defendants"), and Xerox.
- "ERISA" shall mean: the Employee Retirement Income Security Act of 1974, as amended, including all regulations promulgated thereunder, and court decisions interpreting such Act or regulations.
 - "Fairness Hearing" shall have the meaning set forth in Section 2.2.4. 1.13

- 1.14 "Final" shall mean: with respect to any judicial ruling or order, that the period for any appeals, petitions, motions for reconsideration, rehearing, or certiorari or any other proceedings for review ("Review Proceeding") has expired without the initiation of a Review Proceeding, or, if a Review Proceeding has been timely initiated, that there has occurred a full and final disposition of any such Review Proceeding without a reversal or any material modification, including the exhaustion of proceedings in any remand and/or subsequent appeal after remand.
 - 1.15 "Financial Institution" shall have the meaning set forth in Section 7.1.1.
- 1.16 "Independent Fiduciary" shall mean the Person retained for the purposes set forth in Section 2.4.
- "Judgment" shall have the meaning set forth in Section 2.2.4. A proposed form of the *Judgment* is attached hereto as Exhibit B.
- "Named Plaintiffs" shall mean: Plaintiffs David Alliet, Linda Willis, Chervl Wright and Thomas Patti.
 - "Parties" shall mean: the Plaintiffs and the Defendants. 1.19
- "Person" shall mean: an individual, partnership, corporation, governmental entity or any other form of entity or organization.
- 1.21 "Plaintiffs" shall mean: Named Plaintiffs and each member of the Settlement Class.
 - 1.22 "Plaintiff Releasees" shall have the meaning set forth in Section 3.3.
- "Plans" shall mean: The Xerox Corporation Savings Plan and The Savings Plan of Xerox Corporation and the Xerographic Division, UNITE HERE (Formerly "The Savings Plan of Xerox Corporation and the Xerographic Division, Union of Needletrades, Industrial and Textile Employees, A.F.L.-C.I.O.-C.L.C."), together with their predecessors and successors, and any trust created under such plans.
- "Plan of Allocation" shall mean: the plan of allocation approved by the Court. Attached as Exhibit C hereto is the form of plan of allocation that will be submitted to the Court for approval.
 - "Preliminary Approval Order" shall have the meaning set forth in Section 2.2.1. 1.25
 - 1.26 "Preliminary Motion" shall have the meaning set forth in Section 2.2.1.
 - 1.27 "Released Claims" shall have the meaning set forth in Section 3.2.

- "Released Parties" shall mean: the Defendants, the Plans, every Person who, at any time during or after the Class Period, was a director, officer, employee or agent of the Company or a trustee or fiduciary of either or both of the Plans, together with, for each of the foregoing, any predecessors, Successors-In-Interest, present and former Representatives, direct or indirect parents and subsidiaries, insurers and any *Person* that controls, is controlled by, or is under common control with any of the foregoing.
 - 1.29 "Releases" shall mean the releases set forth in Article 3.
- 1.30 "Representatives" shall mean: attorneys, agents, directors, officers, and employees.
- "Securities Actions" shall mean: Carlson v. Xerox Corporation, Case No. 3:00-1.31 CV-1621 (AWT) (D. Conn.), In re: Xerox Securities Litig., 3:99-CV-02374 (AWT) (D. Conn.), and any and all cases now or hereafter consolidated therewith, and the SEC Fair Fund. Securities Actions shall not include the Action.
- "Settlement" shall mean: the settlement to be consummated under this Settlement 1.32 Agreement.
- "Settlement Administrator" shall mean: the Person who is so appointed pursuant 1.33 to the *Preliminary Approval Order* and who is responsible for calculating, implementing, and overseeing the Plan of Allocation.
- "Settlement Class" shall mean: all current and former participants and 1.34 beneficiaries of the Plans for whose individual accounts the Plans purchased and/or held interests in the Xerox Stock Fund at any time during the period May 12, 1997 through and including June 28, 2002, excluding the *Individual Defendants*.
 - "Settlement Fund" shall have the meaning set forth in Section 7.1. 1.35
- "Successor-In-Interest" shall mean: a Person's estate, legal representatives, heirs, successors or assigns, including successors or assigns that result from corporate mergers or other structural changes.

2. Implementation of the *Settlement*.

This Settlement shall be implemented in accordance with the provisions of this Settlement Agreement including Section 2 hereof, and, subject to the provisions of Section 10 hereof, shall be final when the conditions specified in Section 2 have been satisfied or waived.

2 1 Class Certification. The Court shall have certified this Action as a class action for settlement purposes only pursuant to Rule 23(a)(1)-(4), 23(b)(1)(A) and/or (B) and 23(e) of the Federal Rules of Civil Procedure, and with a Settlement Class defined as set forth in Section 1.34 above as a non-opt-out class. The Parties stipulate to certification of the Settlement Class for purposes of this Settlement only. The Parties agree that if the Court does not enter the

Page 14 of 38

Judgment, or if the Judgment does not become Final, no Settlement Class will be deemed to have been certified by or as a result of this Settlement Agreement, and the Action will for all purposes with respect to the Parties revert to its status as of December 15, 2008. In such event the Defendants will not be deemed to have consented to the certification of any class, the agreements and stipulations in this Settlement Agreement concerning class definition or class certification shall not be used as evidence or argument to support class certification or class definition, and the *Defendants* will retain all rights to oppose class certification.

- 2.2 <u>Court Approval</u>. The *Settlement* shall have been approved by the *Court*, as provided for in this Section 2.2. Co-Lead Counsel shall move the Court for an order and judgment approving this Settlement Agreement and the Settlement contemplated hereunder in the form of the Judgment attached as Exhibit B hereto. The Named Plaintiffs shall recommend to the Court that such order and judgment be entered; the Defendants will not object to such recommendation; and the Parties shall cooperate in good faith, including taking all steps and efforts contemplated by this Settlement Agreement and any other steps or efforts that may become necessary by order of the *Court* (unless such order materially modifies the terms of this Settlement Agreement), to carry out this Settlement Agreement, including the following:
- 2.2.1 Motion for Preliminary Approval of Settlement and of Notice. As soon as reasonably possible upon the full execution of this Settlement Agreement by the Parties, Named Plaintiffs will file a motion or other appropriate document(s) as directed by the Court ("Preliminary Motion") with the Court for an order substantially in the form annexed hereto as Exhibit A, including the exhibits thereto (the "Preliminary Approval Order"). The Parties waive oral argument on the Preliminary Motion.
- 2.2.2 Request by Court, Named Plaintiffs, or Defendants for Information. If the Court requests or orders Named Plaintiffs or Defendants to supply non-privileged information in their possession as part of the Court's review of the Settlement Agreement, the Named Plaintiffs and Defendants agree to promptly provide such information. If Named Plaintiffs deem it necessary for the *Defendants* to supply non-privileged information in their possession, and not otherwise available to the Named Plaintiffs, in order to respond to any timely filed objection or Court request/order, the *Defendants* agree to promptly provide such non-privileged information that has been reasonably requested, which shall be treated as Confidential under the Amended Consent Protective Order entered by the Court in this Action on August 1, 2006. If Defendants deem it necessary for the *Named Plaintiffs* to supply non-privileged information in their possession in order to respond to any inquiry from the *Independent Fiduciary* or the Department of Labor, the Named Plaintiffs agree to promptly provide such non-privileged information that has been reasonably requested, which shall be treated as Confidential under the Amended Consent Protective Order entered by the *Court* in this *Action* on August 1, 2006.
- 2.2.3 <u>Issuance of Class Notice</u>. On the date and in the manner set by the Court in its Preliminary Approval Order, the Plaintiffs shall cause the Class Notice to be transmitted in the form and manner approved by the Court as directed in the Preliminary Approval Order. The Parties agree, and the form of Preliminary Approval Order shall provide, that the last known addresses for the *Plaintiffs* in the possession of the *Plans*' current record-keeper will suffice for

all purposes in connection with this *Settlement* and the *Plan of Allocation*, including, without limitation, the mailing of the *Class Notice*.

- 2.2.4 <u>The Fairness Hearing</u>. On or after the date set by the *Court* for the hearing (the "Fairness Hearing") the *Court* will determine: (i) whether to enter judgment finally approving the *Settlement* (which judgment is referred to herein as the "Judgment"); and (ii) the amount of legal fees, compensation, and expenses to be awarded to *Co-Lead Counsel* and *Other Appointed Counsel*, and to the *Named Plaintiffs* as provided in Section 11 of this *Settlement Agreement*.
- 2.3 <u>Finality of Judgment</u>. The Court shall have entered the *Judgment* substantially in the form attached as Exhibit B hereto, as more fully discussed in Section 2.2, and the *Judgment* shall have become *Final*.
- 2.4 <u>Settlement Authorized by Independent Fiduciary</u>. At least twenty (20) days prior to the *Fairness Hearing*, an *Independent Fiduciary* chosen by the *Company*, who shall acknowledge in writing that it is a fiduciary on behalf of the *Plans* with respect to the settlement of this *Action*, shall have approved and authorized in writing the *Settlement* in accordance with Prohibited Transaction Class Exemption 2003-39. If the *Independent Fiduciary* disapproves or otherwise does not authorize the *Settlement*, the *Defendants* shall have the option to waive this condition. Such option is to be exercised in writing within the earlier of (i) ten (10) days of the *Defendants*' receipt of the *Independent Fiduciary*'s written determination, or (ii) three (3) days prior to the date set for the *Fairness Hearing*. If the *Defendants* elect in writing to waive this condition, the *Settlement* shall become *Final* if all other conditions in Section 2 are satisfied. If the *Defendants* elect not to waive this condition, the *Settlement Agreement* shall terminate and become null and void and the provisions of Section 10.2 shall apply.
- 2.5 <u>Assurances From the Department of Labor</u>. *Defendants* may discuss this *Settlement* and the subject matter of this *Action* with the Department of Labor. If the Defendants discuss the *Settlement* or the subject matter of this *Action* with the Department of Labor and the *Defendants* fail to obtain reasonable assurance from the Department of Labor that it will not object to this *Settlement* or institute suit or an administrative proceeding against any *Defendant* at least twenty-five (25) days prior to the *Fairness Hearing*, and if *Xerox*, in its sole discretion, fails to waive this condition, then, upon *Xerox* notifying *Named Plaintiffs* in writing that this condition has failed, the *Settlement Agreement* shall terminate and become null and void and the provisions of Section 10.2 shall apply; provided that this reasonable assurance may be contingent on this *Settlement* becoming *Final*.
- Approval Order, there shall be negotiated with the putative Settlement Administrator the form of agreement, acceptable to Named Plaintiffs and Xerox in their respective reasonable discretion, under which the Settlement Administrator will, subject to its appointment in the Preliminary Approval Order, perform the functions set forth in this Settlement Agreement relating to implementation of the Plan of Allocation. If such form of agreement is not reached prior to entry of the Preliminary Approval Order, then each of Named Plaintiffs and Xerox shall have the right to terminate this Settlement Agreement by written notice to the other, in which event the

Settlement Agreement shall terminate and become null and void and the provisions of Section 10.2 shall apply.

3. Releases.

- Releases of the Released Parties. Subject to Section 10 hereof, and the Class 3.1 Settlement Amount stated in Paragraph 7.2 hereof having been deposited by Defendants into the Settlement Fund, effective upon the entry of the Judgment by the Court, Named Plaintiffs on behalf of themselves and on behalf of the Settlement Class and the Plans absolutely and unconditionally release and forever discharge with prejudice the Released Parties from Released Claims that the Plans, the Plaintiffs, the Settlement Class or Named Plaintiffs directly, indirectly, derivatively, or in any other capacity ever had, now have or hereafter may have, except that the release under this Section 3.1 shall not include claims relating to the covenants or obligations set forth in this Settlement Agreement. Also effective upon entry of the Judgment by the Court, the Plans, the Plans' fiduciaries, the Named Plaintiffs and all other members of the Settlement Class shall be permanently and finally enjoined without the necessity of posting a bond from commencing or prosecuting any actions or other proceedings asserting any of the Released Claims either directly, indirectly, derivatively, or in any other capacity, against any of the Released Parties.
- Released Claims. Subject to Section 3.4 below, the Released Claims shall be: any 3.2 and all claims of any nature whatsoever (including claims for any and all losses, damages, unjust enrichment, attorneys' fees, disgorgement of fees, litigation costs, injunction, declaration, contribution, indemnification or any other type or nature of legal or equitable relief), whether accrued or not, whether known, unknown, or unsuspected, in law or equity, as well as any claim or right obtained by assignment, brought by way of demand, complaint, cross-claim, counterclaim, third-party claim or otherwise (collectively, "Claims"), in any court or other tribunal, arising out of or in any way related to, directly or indirectly, any or all of the acts, omissions, facts, matters, transactions or occurrences, (i) that are, were or could have been alleged, asserted, or set forth in the Complaint or (ii) that would be barred by principles of res judicata had the claims asserted in the Complaint been fully litigated and resulted in a Final judgment or order. With respect to the *Released Claims*, it is the intention of the *Parties* and of the *Plans*, the *Named Plaintiffs* and all other members of the *Settlement Class* expressly to waive to the fullest extent of the law: (a) the provisions, rights and benefits of Section 1542 of the California Civil Code, which provides that "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor"; and (b) the provisions, rights and benefits of any similar statute or common law of any other jurisdiction that may be, or may be asserted to be, applicable. The Released Claims do not include any ERISA Section 502(a)(1)(B) claim for vested benefits by any Plaintiff where such claims are unrelated to any matter asserted in this Action.
- Defendants' Releases of Named Plaintiffs, the Settlement Class and Co-Lead and 3.3 Other Appointed Counsel. Subject to Section 10 hereof, effective upon the entry of the Judgment by the Court, the Defendants absolutely and unconditionally release and forever discharge with prejudice the Named Plaintiffs, the Settlement Class, Co-Lead Counsel, Other

Appointed Counsel, Dalton Gotto Samson & Kilgard PLC and Susman, Heffner & Hurst, LLP, and their and Co-Lead Counsel and Other Appointed Counsel's current and former members, partners, partnerships, professional corporations and firms, and their respective shareholders, partners, officers, directors, associates, agents, employees, consultants, Of Counsels, experts, vendors, predecessors (including all firms with which any Person currently employed or affiliated with Co-Lead Counsel or Other Appointed Counsel was employed or affiliated during the pendency of the Action, and the current and former members, partners, partnerships, shareholders, partners, officers, directors, associates, agents, employees, consultants, and Of Counsels of such firms), successors and assigns (collectively, the "Plaintiff Releasees"), from any and all *Claims* relating to the institution or prosecution of the *Action* or the settlement of any Released Claims, except that the release under this Section 3.3 shall not include claims relating to the covenants or obligations set forth in this Settlement Agreement or in the Amended Consent Protective Order entered by the *Court* in this *Action* on August 1, 2006.

<u>Claims Not Affected.</u> Nothing in this Settlement Agreement shall affect the settlement and release of claims in the Securities Actions, or bar, waive or otherwise affect any Claim that has been or could be asserted under state or federal securities laws by the Plans and/or any member of the Settlement Class in the Securities Actions.

4. Covenants.

The *Parties* covenant and agree as follows:

- Taxation of Class Settlement Amount. The Released Parties shall have no 4.1 responsibility for any taxes due with respect to funds deposited in, income earned on or funds distributed from the Settlement Fund, including funds that the Plaintiffs or Co-Lead Counsel or Other Appointed Counsel receive from the Class Settlement Amount.
- Cooperation. The Company shall cooperate with Co-Lead Counsel by providing, 4.2 in electronic format, the names and addresses of *Persons* to whom the *Class Notice* is to be sent pursuant to the Preliminary Approval Order, in accordance with Section 2.2.3. Plaintiffs shall use any information provided by the *Company* pursuant to the preceding sentence solely for the purpose of providing notice in this *Action* and for no other purpose. The *Parties* shall reasonably cooperate with each other to effectuate this Settlement, and shall not do anything or take any position inconsistent with obtaining a prompt Judgment approving the Settlement unless expressly permitted by this Settlement Agreement.
- Covenant Not to Sue. Subject to Section 10 hereof, *Plaintiffs* covenant and agree 4.3 on their own behalf, and on behalf of the Settlement Class and the Plans: (i) not to file against any Released Party any Claim based on, related to, or arising from any Released Claim; and (ii) that the foregoing covenants, agreements and releases shall be a complete defense to any such Claims against any of the respective Released Parties.
- Confidentiality. Except as otherwise provided by law or as may result in the ordinary course from submission of the Settlement to the Court pursuant to Section 2.2.1 hereof, the Parties agree that the Settlement shall remain confidential until the Company publicly

announces the Settlement. Co-Lead Counsel and Other Appointed Counsel agree to refrain from publishing, releasing, or otherwise publicly communicating any terms of the Settlement in advance of the public announcement of the Settlement by the Company without the prior approval of the Company acting on behalf of Defendants. Defendants retain the right to disclose the existence and terms of this *Settlement* to their external auditors and insurers at any time. Named Plaintiffs, on the one hand, and Defendants, on the other, further agree that they shall not disparage the other in any public statements regarding the Action, the Settlement, or the facts the subject matter of the Action. As used herein, "public statement" means any statement made in the press, broadcast, cable, or electronic media, or otherwise made in a manner designed to make the statement available to the public generally.

5. Representations and Warranties.

5.1 Named Plaintiffs' Representations and Warranties.

- 5.1.1 Named Plaintiffs represent and warrant that they have not assigned or otherwise transferred any interest in any Released Claims against any Released Party, and further covenant that they will not assign or otherwise transfer any interest in any Released Claims: and
- 5.1.2 Pursuant to Sections 3 and 4, *Named Plaintiffs* represent and warrant that upon this Settlement becoming final, they shall have no Released Claims against any of the Released Parties.
- Parties' Representations and Warranties. The Parties, and each of them, 5.2 represent and warrant as follows, each acknowledging that each is relying on these representations and warranties in entering into this Settlement Agreement:
- 5.2.1 That they have conducted voluminous discovery and have diligently prepared for trial pursuant to the Court's orders; that they are voluntarily entering into this Settlement Agreement as a result of arm's-length negotiations among their counsel, with the assistance of an experienced mediator; that in executing this Settlement Agreement they are relying solely upon their own judgment, belief and knowledge, and the advice and recommendations of their own independently selected counsel concerning the nature, extent and duration of their rights and claims hereunder and regarding all matters that relate in any way to the subject matter hereof; and that, except as otherwise explicitly set forth in this Settlement Agreement, they have not been influenced to any extent whatsoever in executing this Settlement Agreement by any representations, statements, or omissions pertaining to any of the foregoing matters by any Party or by any Person representing any Party to this Settlement Agreement. Each *Party* assumes the risk of mistake as to facts or law, except for the express representations and warranties set forth in this Settlement Agreement.
- 5.2.2 That they have carefully read the contents of this Settlement Agreement, and this Settlement Agreement is signed freely by each Person executing this Settlement Agreement on behalf of each of the Parties. The Parties, and each of them, further represent and warrant to each other that he, she, or it has made such investigation of the facts pertaining to the

Settlement, this Settlement Agreement, and all of the matters pertaining thereto, as he, she, or it deems necessary.

- 5.2.3 That in entering into this Settlement Agreement, the Plaintiffs have not relied on any representations or arguments by the *Defendants* that a class can, could, or should be certified for any purpose; that the Plaintiffs would not be prejudiced if (i) this Settlement were not approved or such approval were reversed on appeal and (ii) the Defendants were later to object to the certification of any proposed class in this action; and that in the event that this Settlement does not become Final, the Plaintiffs will not assert that the Defendants are equitably or judicially estopped from contesting the certification of any class in this *Action*.
- Signatories' Representations and Warranties. Each Person executing this 5.3 Settlement Agreement on behalf of any other Person does hereby personally represent and warrant to the other Parties that he or she has the authority to execute this Settlement Agreement on behalf of, and fully bind, each principal whom such individual represents or purports to represent.

6. No Admission of Liability.

The Parties understand and agree that this Settlement Agreement embodies a compromise settlement of disputed claims, and that nothing in this Settlement Agreement, including the furnishing of consideration for this Settlement Agreement, shall be deemed to constitute any finding of fiduciary status under ERISA or wrongdoing by any of the Defendants, or give rise to any inference of fiduciary status under ERISA or wrongdoing or admission of wrongdoing or liability in this or any other proceeding. This Settlement Agreement and the payments made hereunder are made in compromise of disputed claims and are not admissions of any liability of any kind, whether legal or factual. The Defendants specifically deny any such liability or wrongdoing and state that they are entering into this Settlement Agreement to eliminate the burden and expense of further litigation. Further, the Named Plaintiffs, while believing that all Claims brought in the Action have merit, have concluded that the terms of this Settlement Agreement are fair, reasonable and more than adequate to the Plans, themselves and members of the Settlement Class given, among other things, the inherent risks, difficulties and delays in complex ERISA litigation such as this. Neither the fact nor the terms of this Settlement Agreement shall be used or offered or received in evidence in any action or proceeding for any purpose, except in an action or proceeding to enforce this Settlement Agreement or arising out of or relating to the *Judgment*.

7. The Settlement Fund; Deliveries into the Settlement Fund.

7.1 The Settlement Fund.

7.1.1 Within five (5) days after the Agreement Execution Date, Co-Lead Counsel shall establish at a federally-insured financial institution (the "Financial Institution") identified to and reasonably agreed on by counsel for Xerox, a settlement fund account (the "Settlement Fund") which, along with net earnings thereon, shall be considered a common fund created as a result of the Action. Each Co-Lead Counsel firm and counsel for Xerox shall

designate at least one person with signature authority over this account (the "Signers"), and shall direct the Financial Institution to make distributions from the Settlement Fund only in accordance with this Settlement Agreement upon written direction from each Signer; provided that the Parties agree that upon the Judgment becoming Final, the Signers will notify the Financial Institution that written direction from counsel for Xerox shall no longer be required. For the avoidance of doubt, the Financial Institution shall be instructed that, absent a Court order, no funds are to be paid or withdrawn from the Settlement Fund except pursuant to Section 8 of this Settlement Agreement (and the Sections of this Settlement Agreement explicitly crossreferenced therein) or, upon termination of this Settlement Agreement, pursuant to Section 10 of this Settlement Agreement. Co-Lead Counsel shall promptly notify Xerox of the date of the establishment of the Settlement Fund, shall confirm the identity of the Financial Institution including any information, including but not limited to wiring instructions, needed to make the deposit in section 7.2, and shall confirm that withdrawals and distributions from the Settlement *Fund* are subject to the restrictions set forth in the preceding sentence.

7.1.2 The funds on deposit in the Settlement Fund shall be invested only in United States Treasury securities and/or securities of United States agencies backed by the full faith and credit of the United States Treasury, and mutual funds or money market accounts that invest exclusively in the foregoing securities. The Settlement Fund shall be structured and managed to qualify as a "qualified settlement fund" described in the Treasury regulations promulgated under Section 468B of the Internal Revenue Code and no Party shall take any position in any filing or before any tax authority that is inconsistent with such treatment. The Financial Institution or another person designated by Co-Lead Counsel (other than a Released Party) shall be the Settlement Fund "administrator," as that term is used in the Section 468B Treasury regulations (the "Administrator"). The Administrator shall (a) prepare and file all income tax and information returns required to be filed, and provide pavees with copies of such information returns; (b) pay all taxes owed by the Settlement Fund; (c) at the Company's request, join with Xerox in timely making the "relation-back" election permitted under the Section 468B Treasury regulations in the form prescribed therein; (d) pay the fees and expenses incurred by the Financial Institution associated with the administration of the Settlement Fund and (e) obtain and provide the Company with the Settlement Fund's federal taxpayer identification number on or before the date that the Company transfers funds to the Settlement Fund. The Administrator shall be authorized to retain a certified public accounting firm for these purposes. All taxes on the income of the Settlement Fund and tax-related expenses, including the expenses, if any, of a certified public accounting firm, incurred in connection with the administration of the Settlement Fund shall be paid solely out of the Settlement Fund, shall be considered a cost of administration of the Settlement, and shall be timely paid without further order of the Court. All fees and expenses of the Administrator or the Financial Institution, and of professional advisors engaged by the Administrator or the Financial Institution in connection with the Settlement Fund, shall be funded solely from the Settlement Fund. Co-Lead Counsel, with written authority from the Signers, may instruct the Financial Institution to reserve any portion of the Settlement Fund for the purpose of satisfying future or contingent expenses or obligations, including expenses of Settlement Fund administration or any disbursement provided in Section 8 of this Settlement The Parties agree that neither Defendants nor any Released Party has any responsibility, authority, or liability, respecting the operation, expenses, obligations and administration of the Settlement Fund

- 7.2 The Class Settlement Amount. In consideration of, and expressly in exchange for, all of the promises and agreements set forth in this Settlement Agreement, the Company will cause to be deposited into the Settlement Fund within ten (10) business days after the entry of the Preliminary Approval Order or after the receipt of the wiring instructions and Financial Institution information referenced in Section 7.1.1, whichever is later, the aggregate sum of fiftyone million dollars in United States currency (\$51,000,000.00) (the "Class Settlement Amount").
- Sole Monetary Contribution. The Class Settlement Amount shall be the full and 7.3 sole monetary contribution made by or on behalf of the Defendants in connection with the Settlement effected between Plaintiffs and the Defendants under this Settlement Agreement. Provided the Class Settlement Amount is deposited in the Settlement Fund as provided in Section 7.1, no *Individual Defendant* will have any personal obligation to fund any or all of the *Class* Settlement Amount.

8. Payments From The Settlement Fund.

- Expenses of Class Notice. After the entry of the Preliminary Approval Order, Co-Lead Counsel shall direct the Financial Institution in writing to disburse from the Settlement Fund an amount for the payment of reasonable, customary and feasible costs associated with the Class Notice, not to exceed \$200,000. If the Settlement Agreement is terminated for any reason, neither Co-Lead Counsel nor any other person shall have any obligation to reimburse to the Settlement Fund the costs of the Class Notice, or any other reasonable costs or expenses of the Settlement Fund charged to, or paid or incurred by or out of the Settlement Fund under this Settlement Agreement.
- <u>Disbursements from Settlement Fund</u>. Except as provided in Section 7.1.2, Section 8.1, or in Section 10 below, no distribution of any part or all of the Settlement Fund shall be paid from the Settlement Fund until the Financial Institution has received (a) a joint notice signed by Co-Lead Counsel and by counsel for the Company, or (b) a Court order directing that the Settlement Fund be disbursed. When the condition in (a) or (b) has been satisfied, Co-Lead Counsel shall direct the Financial Institution to disburse money from the Settlement Fund as follows:
 - For Attorneys' Fees and Expenses. As provided in Section 11.2 hereof. 8.2.1
 - <u>For Named Plaintiff compensation</u>. As provided in Section 11.2 hereof. 8.2.2
- 8.2.3 For taxes and expenses of the Settlement Fund. As provided in Section 7.1.2 hereof.
- 8.2.4 For Settlement Administrator Expenses. Settlement Administrator Expenses shall mean all reasonable fees and expenses of the Settlement Administrator to calculate, implement, and oversee the *Plan of Allocation*, including, but not limited to, expenses incurred in calculating and allocating the share of the Net Proceeds, as defined in Section 8.2.5, for each member of the Settlement Class. The Settlement Fund (and this allocation) shall be the

sole source used to pay for these fees and expenses. The *Parties* agree to cooperate to seek to minimize the *Settlement Administrator Expenses*. The *Parties* also agree that *Settlement Administrator Expenses* shall not be incurred until reasonably necessary in light of the conditions to the *Settlement*.

8.2.5 For the *Plan of Allocation*. The *Plan of Allocation* attached as Exhibit C hereto provides for the allocation of the Settlement Fund net of the disbursements called for in Sections 8.2.1, 8.2.2, 8.2.3, and 8.2.4 ("Net Proceeds"). Upon the Judgment becoming Final as provided in Section 2.3, and after the amounts payable pursuant to Sections 8.2.1, 8.2.2, 8.2.3 and 8.2.4 have been determined and disbursed, Co-Lead Counsel shall direct the Financial Institution to disburse the Net Proceeds to the master trust for the Plans for allocation to or for the benefit of members of the Settlement Class. The Parties agree that the deposit of the Net Proceeds into the trust created under the Plans shall constitute "restorative payments" within the meaning of Revenue Ruling 2002-45 for all purposes. Defendants shall direct State Street Bank and Trust Company, or any successor Trustee of the *Plans*, or any other authorized entity, to allocate the Net Proceeds received by the Plans' trust according to the Plan of Allocation and shall notify Co-Lead Counsel as to the date(s) and amount(s) of said allocation(s), but Defendants shall have no liability for the failure by State Street Bank and Trust Company, or any successor Trustee of the *Plans*, the *Plans*' record-keeper, or any other authorized entity, to follow such directions provided that, in the event of any failure by State Street Bank and Trust Company, or any successor Trustee of the Plans, or any other authorized entity, to follow directions from Defendants given pursuant to this Section 8.2.5, the Defendants shall assist in seeking to enforce such directions. All reasonable fees and expenses of State Street Bank and Trust Company, or any predecessor or successor Trustee of the *Plans*, the *Plans*' record-keeper or any other authorized entity, with respect to implementation of the *Plan of Allocation* shall be Plans' expenses paid solely out of the Net Proceeds and shall be timely paid by the Plans without further order of the Court. Defendants warrant that they either have obtained or will obtain the authority to direct that the Net Proceeds received by the Plans' trust be allocated according to the Plan of Allocation. Defendants shall have no responsibility for structuring the content of the Plan of Allocation, or for its design or implementation, but will have the right to review it for feasibility and cost before presentation to the Court. The Plan of Allocation is a matter separate and apart from the Settlement between the Parties, and no decision by the Court that modifies the Plan of Allocation shall in any manner affect the validity of this Settlement Agreement, or the finality of the Settlement, or the binding character of the releases provided for hereunder.

9. Additional Conditions

9.1 <u>Asset Management</u>. The Xerox Retirement Investment Committee ("XRIC"), consisting of specified members of the *Company*'s management (Treasurer; Chief Accounting Officer; Vice President, Worldwide Taxes; Vice President, Human Resources; and Chief Information Officer), will abide by the terms of each of the *Plans* with respect to asset management, including asset management of the Xerox Stock Fund, in a manner consistent with *ERISA* and other applicable law.

- 9.2 Other Funds. Unless otherwise required by law, the *Company* shall not amend the Plans to prevent participants and beneficiaries from investing all or any part of their account balances in funds other than the Xerox Stock Fund.
- XRIC Composition. The composition of XRIC will be identified to participants in 9.3 the *Plans*.
- 9.4 <u>Disclosure</u>. Annually, the investment objectives of the Xerox Stock Fund will be published to the participants and beneficiaries of each of the *Plans*.
- 9.5 Independent Fiduciary. The Company shall pay the costs of the Independent Fiduciary using corporate assets.

10. Termination of the Settlement Agreement.

- 10.1 Termination. This Settlement Agreement may terminate if (a) the Court declines to enter the *Judgment*, (b) the *Judgment* entered by the *Court* is reversed or modified in any material respect on appeal, (c) the *Independent Fiduciary* does not approve the *Settlement* as set forth in Section 2.4, and such condition is not waived in writing by Defendants, or (d) the Department of Labor takes any of the actions described in Section 10.3. This Settlement Agreement shall not terminate if a court of competent jurisdiction modifies, reverses, or refuses to enter any order relating to the award of legal fees and expenses, or Named Plaintiff compensation, or the Plan of Allocation. If, within thirty (30) days after the date when any reversal or modification which would cause this Settlement Agreement to terminate becomes Final, the Parties have not agreed in writing to proceed with all or part of the Settlement Agreement in light of such ruling, this Settlement Agreement shall automatically terminate and thereupon become null and void.
- Consequences of Termination of the Settlement Agreement. If the Settlement 10 2 Agreement terminates, the following shall occur:
- 10.2.1 Co-Lead Counsel and Defendants' Counsel shall within ten (10) days after the date of termination of the Settlement Agreement jointly notify the Financial Institution in writing to return to the *Defendants* the amount contributed by them to the *Settlement Fund*, with all net income earned thereon, after deduction of the amount earlier disbursed for Class Notice described in Section 8.1 and the expenses charged by the Administrator and the Financial Institution, directing the Financial Institution to effect such return within fourteen (14) days after such notification. Prior to the return of amounts contemplated by this Section 10.2.1, the Administrator shall fully and finally fulfill all tax obligations of the Settlement Fund as set forth in Section 7.1.2 and the *Defendants* shall have no past, present, or future liability whatsoever for any such tax obligations.
- 10.2.2 The Action shall for all purposes with respect to the Parties revert to its status as of December 15, 2008. Any and all statutes of limitations, statutes of repose and/or other defenses based upon the passage of time applicable to the Claims asserted in this Action

shall be tolled from December 15, 2008 to the date on which this Settlement Agreement terminates.

- 10.2.3 All provisions of this *Settlement Agreement* shall be null and void except as otherwise explicitly provided in this *Settlement Agreement*.
- 10.3 <u>Intervention by the Department of Labor</u>. If, prior to the entry of the *Judgment* in this case, the Department of Labor files an objection to the *Settlement*, files an amicus curiae brief disapproving of or opposing the *Parties' Settlement* in this *Action* or files its own lawsuit against any of the *Defendants* asserting allegations and claims similar to those asserted in the *Action, Xerox* shall have the right to terminate this *Settlement Agreement* at its sole discretion.

11. <u>Attorneys' Fees and Expenses and Named Plaintiff Compensation.</u>

- 11.1 <u>Application for Attorneys' Fees and Expenses and Named Plaintiff</u> <u>Compensation</u>. As provided in Section 2.2, and pursuant to the common fund doctrine and/or any applicable statutory fee provision, *Co-Lead Counsel* may apply to the *Court* for an award to *Co-Lead Counsel* and *Other Appointed Counsel* of attorneys' fees, and for reimbursement of expenses, to be paid solely from the *Settlement Fund*. *Co-Lead Counsel* also may apply to the *Court* for compensation to *Named Plaintiffs*, payable solely from the *Settlement Fund*, and *Named Plaintiffs* shall be entitled to receive such compensation from the *Settlement Fund* to the extent awarded by the *Court*.
- 11.2 <u>Disbursement of Attorneys' Fees and Expenses and Named Plaintiff</u> Compensation. If the Court enters one or more orders allowing payment of attorneys' fees and/or expenses and/or Named Plaintiff compensation, then Co-Lead Counsel may, not less than five (5) days after the Judgment becomes Final, instruct the Financial Institution in writing to disburse the payments set forth in such order(s) from the Settlement Fund.

12. Miscellaneous Provisions.

- 12.1 <u>Governing Law</u>. This *Settlement Agreement* shall be governed by the laws of the State of New York without giving effect to the conflict of laws or choice of law provisions thereof, except to the extent that the law of the United States governs any matter set forth herein, in which case such federal law shall govern.
- 12.2 <u>Amendment</u>. Before entry of the *Judgment*, the *Settlement Agreement* may be modified or amended only by written agreement signed by or on behalf of all *Parties*. Following entry of the *Judgment*, the *Settlement Agreement* may be modified or amended only by written agreement signed by or on behalf of all *Parties* and approved by the *Court*.
- 12.3 <u>Waiver</u>. The provisions of this *Settlement Agreement* may be waived only by an instrument in writing executed by the waiving party. The waiver by any party of any breach of this *Settlement Agreement* shall not be deemed to be or construed as a waiver of any other breach, whether prior, subsequent, or contemporaneous, of this *Settlement Agreement*.

Page 25 of 38

- Construction. None of the Parties hereto shall be considered to be the drafter of this Settlement Agreement or any provision hereof for the purpose of any statute, case law, or rule of interpretation or construction that would or might cause any provision to be construed against the drafter hereof.
- Principles of Interpretation. The following principles of interpretation apply to 12.5 this Settlement Agreement:
- 12.5.1 Headings. The headings of this Settlement Agreement are for reference purposes only and do not affect in any way the meaning or interpretation of this Settlement Agreement.
- 12.5.2 Singular and Plural. Definitions apply to the singular and plural forms of each term defined.
- 12.5.3 Gender. Definitions apply to the masculine, feminine, and neuter genders of each term defined.
- 12.5.4 References to a Person. References to a Person are also to the Person's permitted successors and assigns.
- 12.5.5 Terms of Inclusion. Whenever the words "include," "includes," or "including" are used in this Settlement Agreement, they shall not be limiting but rather shall be deemed to be followed by the words "without limitation."
- 12.5.6 Time. References to "days" in this Settlement Agreement are to calendar days, unless otherwise stated.
- <u>Further Assurances</u>. All *Parties* agree, without further consideration, and as part of finalizing the *Settlement* hereunder, that they will in good faith execute and deliver such other documents and take such other actions as may be necessary to consummate and effectuate the subject matter and purpose of this Settlement Agreement.
- Notices. Any notice, demand, or other communication under this Settlement 12.7 Agreement (other than the Class Notice, or other notice given at the direction of the Court) shall be in writing and shall be deemed duly given upon receipt if it is addressed to each of the intended recipients as set forth below and personally delivered, sent by registered or certified mail (postage prepaid), sent by confirmed facsimile, or delivered by reputable express overnight courier:

IF TO NAMED PLAINTIFFS:

Lynn Sarko Gary Gotto Elizabeth Leland Keller Rohrback L.L.P.

1201 Third Avenue, Suite 3200 Seattle, WA 98101 Telephone: (206) 623-1900

Facsimile: (206) 623-3384

Charles Watkins John Wylie Futterman Howard Watkins Wylie & Ashley, Chtd 122 S. Michigan Avenue, Suite 1850 Chicago, Illinois 60603 Telephone: (312) 427-3600

Facsimile: (312) 427-1850

IF TO DEFENDANTS

Steven Sacher Evan Miller Jones Day 51 Louisiana Ave. N.W. Washington, DC 20001-2113 Telephone: (202) 879-3939 Facsimile: (202) 626-1700

Any Party may change the address at which it is to receive notice by written notice delivered to the other *Parties* in the manner described above.

- Entire Agreement. This Settlement Agreement contains the entire agreement among the *Parties* relating to this *Settlement*. It specifically supersedes any settlement terms or settlement agreements relating to the Defendants that were previously agreed upon orally or in writing by any of the *Parties*. In the event of any dispute regarding the interpretation of this Settlement Agreement, no Party may rely upon or cite any previous version of this Settlement Agreement.
- Counterparts. This Settlement Agreement may be executed by exchange of faxed or scanned executed signature pages, and any signature thereby transmitted for the purpose of executing this Settlement Agreement shall be deemed an original signature for purposes of this Settlement Agreement. This Settlement Agreement may be executed in two or more counterparts, each of which shall be deemed to be an original, but all of which, taken together, shall constitute one and the same instrument.
- 12.10 Binding Effect. This Settlement Agreement binds and inures to the benefit of the Parties hereto, their assigns, heirs, administrators, executors, and successors.
- 12.11 Agreement Execution Date. The date on which the final signature is affixed below shall be the Agreement Execution Date.

12.12 <u>Return of Discovery Documents</u>. Within sixty (60) days after the *Judgment* becomes *Final*, the parties shall comply with the provisions of the Amended Consent Protective Order, issued by the *Court* in this *Action* on August 1, 2006.

IN WITNESS WHEREOF, the Parties have executed this Settlement Agreement on the dates set forth below.

FOR THE PLAINTIFES:

By:

Dated: 1/26/09

Lynn Sarko

Gary Gotto

Elizabeth Leland Keller Rohrback L.L.P.

1201 Third Avenue, Suite 3200

Seattle, WA 98101

Telephone: (206) 623-1900 Facsimile: (206) 623-3384

By:

Dated

Charles Watkins John Wylie

Futterman Howard Watkins Wylie & Ashley, Chtd

122 S. Michigan Avenue, Suite 1850

Chicago, Illinois 60603 Telephone: (312) 427-3600 Facsimile: (312) 427-1850

Co-Lead Counsel for the ERISA Plaintiffs

FOR DEFENDANTS XEROX CORPORATION, LAWRENCE BECKER, SALLY CONKRIGHT, PATRICIA M. NAZEMETZ, ARLYN B. KASTER, KATHLEEN RUSSELL, WILLIAM STRUSZ, LANCE DAVIS, CHRISTINA CLAYTON, GARY KABURECK, LAWRENCE ZIMMERMAN, ANNE M. MULCAHY, B.R. INMAN, VERNON E. JORDAN, JR., HILMAR KOPPER, GEORGE J. MITCHELL, N.J. NICHOLAS, JR., PATRICIA F. RUSSO, MARTHA R. SEGER, THOMAS C. THEOBALD, G. RICHARD THOMAN, BARBARA D. ROSCOE AS BENEFICIARY OF PROPERTY PASSING PURSUANT TO THE LAST WILL AND TESTAMENT OF WILLIAM C. ROSCOE, DATED DECEMBER 30, 2005, HENRY CHARLES FILTER, III. AS BENEFICIARY OF PROPERTY PASSING PURSUANT TO THE WILL OF EUNICE M. FILTER DATED FEBRUARY 25, 2002 TO THE EUNICE M. FILTER REVOCABLE TRUST DATED FEBRUARY 25. 2002, HENRY CHARLES FILTER, III, AND JOHN MUSICARO, JR., AS TRUSTEES OF THE TRUST FOR THE BENEFIT OF HENRY CHARLES FILTER, III AND HIS DESCENDANTS UNDER ARTICLE II OF THE EUNICE M. FILTER INSURANCE TRUST UNDER AGREEMENT DATED JANUARY 19, 1995, AS BENEFICIARY OF PROPERTY PASSING PURSUANT TO THE WILL OF EUNICE M. FILTER DATED FEBRUARY 25, 2002 TO THE EUNICE M. FILTER REVOCABLE TRUST DATED FEBRUARY 25, 2002, C/O JOHN MUSICARO, JR., CUMMINGS & LOCKWOOD LLC, SIX LANDMARK SQUARE, STAMFORD, CT 06901, AND JERRY W. HOSTETTER AND JOHN MUSICARO, JR. AS TRUSTEES OF THE MARITAL TRUST UNDER ARTICLE IV OF THE EUNICE M. FILTER REVOCABLE TRUST DATED FEBRUARY 25, 2002, AS BENEFICIARY OF PROPERTY PASSING PURSUANT TO THE WILL OF EUNICE M. FILTER DATED FEBRUARY 25, 2002 TO THE EUNICE M. FILTER REVOCABLE TRUST DATED FEBRUARY 25. 2002, C/O JOHN MUSICARO, JR., CUMMINGS & LOCKWOOD LLC. SIX LANDMARK SQUARE, STAMFORD, CT:

By:

Steven J. Sacher Evan Miller

Jones Day

51 Louisiana Ave. N.W.

Washington, DC 20001-2113 Telephone: (202) 879-3939

Facsimile: (202) 626-1700

Attorneys for the Defendants except Myra R. Drucker, Paul A. Allaire, William F. Buehler, Barry D. Romeril, and Gregory B. Tayler

FOR DEFENDANT MYRA R. DRUCKER:

By:	Ble T. Hangles Dated: January 26, 2009
-	Michael T. Hannafan
	Blake T. Hannafan
•	Hannafan & Hannafan, Ltd.
	One East Wacker Drive, Suite 2800
	Chicago, IL 60601
	Telephone: (312) 527-0055
	Attorneys for Defendant Myra R. Dručker
EOD	DEFENDANTS PAUL A. ALLAIRE, WILLIAM F. BUEHLER,
POR BAR	RY D. ROMERIL, AND GREGORY B. TAYLER:
<i>101</i> 110	KT D. ROWLAND, The Grant D. L.
	- ·
Ву:	Dated:
	John A. Valentine
	Wilmer Cutler Pickering Hale and Dorr LLP 1875 Pennsylvania Ave., NW
	Washington, DC 20006
•	Telephone: (202) 663-6000
	Terephone. (202) 003-0000
	Attorneys for Defendants Paul A. Allaire, William F. Buehler,
	Barry D. Romeril, and Gregory B. Tayler
DOD	POPER TO ANIE OF DIGITADE TRIOMANI.
FOR	DEFENDANT G. RICHARD THOMAN:
D	Dated:
By:	
	Thomas D. Goldberg Day Pitney LLP
	One Canterbury Green
	Stamford, CT 06901
	Telephone: (203) 977-7300
	1 Clephone. (203) 311-1300

Attorneys for Defendant G. Richard Thoman

Filed 01/26/2009

Attorneys for the Defendants except Myra R. Drucker, Paul A. Allaire, William F. Buehler, Barry D. Romeril, and Gregory B. Tayler

FOR DEFENDANT MYRA R. DRUCKER:

By:	Dated:
	Michael T. Hannafan Blake T. Hannafan Hannafan & Hannafan, Ltd. One East Wacker Drive, Suite 2800 Chicago, IL 60601 Telephone: (312) 527-0055
	Attorneys for Defendant Myra R. Drucker
BARR	DEFENDANTS PAUL A. ALLAIRE, WILLIAM F. BUEHLER, LY D. ROMERIL, AND GREGORY B. TAYLER:
By:	John A. Valentine Wilmer Cutler Pickering Hale and Dorr LLP 1875 Pennsylvania Ave., NW Washington, DC 20006 Telephone: (202) 663-6000 Attorneys for Defendants Paul A. Allaire, William F. Buehler, Barry D. Romeril, and Gregory B. Tayler
FOR I	DEFENDANT G. RICHARD THOMAN:
By:	Thomas D. Goldberg Day Pitney LLP One Canterbury Green Stamford, CT 06901 Telephone: (203) 977-7300 Attorneys for Defendant G. Richard Thoman

Attorneys for the Defendants except Myra R. Drucker, Paul A. Allaire, William F. Buehler, Barry D. Romeril, and Gregory B. Tayler

1	F	1	P	Г	11	71	71	77	J	Γ	۸	1	ď	Т	١	1	V	D	۸	τ)	Γ	D	T I	0	V	EI	Э.
н	г.		'n		"	7.1	٦.	٦, ١	N		1	N١	N	-	-13	/ I	Y	ĸ	1	1	•		ж			_		1

By:		Dated:									
Dy.	Michael T.										
	Blake T. H										
		& Hannafan, Ltd.									
		Vacker Drive, Suite 2800									
	Chicago, II										
	Telephone: (312) 527-0055										
	Attorneys f	or Defendant Myra R. Drucker									
		TS PAUL A. ALLAIRE, WILLIAM F. BUEHLER, ERIL, AND GREGORY B. TAYLER:									
By:		Dated:									
2).	John A. Va										
	THE RESERVE THE PROPERTY OF TH	tler Pickering Hale and Dorr LLP									
		sylvania Ave., NW									
		n, DC 20006									
		(202) 663-6000									
	relephone.	(202) 003-0000									
		or Defendants Paul A. Allaire, William F. Buehler, omeril, and Gregory B. Tayler									
FOR E	DEFENDAN	T G. RICHARD THOMAN:									
By:	15	0 Dld Dated: 1-22-09									
	Thomas D.	Goldberg									
	Day Pitney	LLP									
	One Canter	bury Green									
	Stamford, C	CT 06901									
		(203) 977-7300									
		2 S									
	Attorneys fo	or Defendant G. Richard Thoman									

EXHIBITS TO THE SETTLEMENT AGREEMENT

Exhibits

A Preliminary Approval Order

B Judgment

C Plan of Allocation

EXHIBIT 2 TO ORDER AND FINAL JUDGMENT

(PLAN OF ALLOCATION)

I. Definitions

- A. All capitalized terms shall have the same meaning as they are given in the Settlement Agreement dated ______ ("Settlement Agreement"), unless specifically set forth otherwise in this Plan of Allocation. The following definitions shall apply to the Plan of Allocation methodology:
- 1. **"Plan Account"** means the account maintained on behalf of a participant in either of the Plans.
- 2. "Member" shall mean a member of the Settlement Class who is or was a Participant and whose Plan Account held any units in the Xerox Stock Fund during the Allocation Period.
- 3. "Current Plan Participant" shall mean: (i) a Member who as of the Reference Date has not received a complete distribution from a Plan; and (ii) a beneficiary or beneficiaries of such a Member.
- 4. "Allocation Period" means, for each Member, the period through and including (a) the earlier of (i) January 2, 1998, or (ii) such date from May 12, 1997 through January 1, 1998, as to which such Member provides to the Settlement Administrator "Adequate 1997 Documentation", and (b) June 28, 2002.
- 5. "Adequate 1997 Documentation" means documentation provided by a Member to the Plan Administrator no later than sixty (60) days after the commencement of the Fairness Hearing, which documentation (i) was created by either of the Plans or their record-keeper or Trustee, or was otherwise contemporaneously created, (ii) shows the acquisition, holding or disposition of units in the Xerox Stock Fund by in such Member's Plan Account

Page 35 of 38

during the period from May 12, 1997 through January 1, 1998, and (iii) is determined by the Settlement Administrator in its discretion to provide sufficiently reliable information for the Settlement Administrator to be able to use it in calculating such Member's Net Loss in accordance with the provisions of Section II of this Plan of Allocation.

- 6. "Former Plan Participant" shall mean (i) a Member who, as of the Reference Date, is not a Current Plan Participant; and (ii) a beneficiary or beneficiaries of such a Member.
- 7. "Xerox Stock" means the common stock of Xerox Corporation ("Xerox") traded on the New York Stock Exchange under the symbol XRX.
- 8. "Participant" means a person who was a participant within the meaning of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), in a Plan during the Allocation Period.
- 9. "Unlocatable Member" shall mean a Member who, despite reasonable efforts, cannot be located by the Plan Administrator within a reasonable time.
- 10. "Reference Date" means the date after the Judgment becomes final, established by the Settlement Administrator for determination of whether a Participant is a Current Plan Participant.

II. Calculation of Allocation

As soon as administratively feasible after the Settlement becomes final in A. accordance with the terms of the Settlement Agreement, the Settlement Administrator shall calculate the share of the Net Proceeds for each Member according to the following methodology:

- B. For each Member, the Settlement Administrator shall determine a net loss ("Net Loss") as follows:
 - 1. Net Loss =(A+B)-(C+D), where, for each Member's Plan Account:
- 2. A = the total dollar value of his or her Plan Account balance in the Xerox Stock Fund at the beginning of the Allocation Period;
- 3. B = the dollar value, if any, of all subsequent investments of any nature whatsoever of such Plan Account in units in the Xerox Stock Fund during the Allocation Period, valued as of the time of each such investment as reflected in the records of a Plan;
- 4. C = the dollar amount of all dispositions of any nature whatsoever of units in the Xerox Stock Fund in the Plan Account, if any, during the Allocation Period, valued as of the time of the disposition, as reflected in the records of a Plan; and
- 5. D = the dollar value of the Plan Account invested in the Xerox Stock Fund at the end of the Allocation Period.
- C. The Settlement Administrator will then total the Net Losses of all Members as calculated in Section II.B to yield the "Aggregate Net Loss," and shall then divide each Member's Net Loss by the Aggregate Net Loss to yield each Member's "Preliminary Fractional Share."
- D. The Settlement Administrator shall then calculate for each Member his or her "Preliminary Dollar Recovery" of the Net Proceeds by multiplying the Member's Preliminary Fractional Share by the Net Proceeds.
- E. The Settlement Administrator shall then identify all Members whose Preliminary Dollar Recovery is less than fifteen dollars (\$15.00) (the "De Minimis Amount"). All such Members' losses shall be deemed to be zero and no allocation will be made to such Members

3

(the "De Minimis Loss Members"). The aggregate Net Losses of the De Minimis Loss Members shall be subtracted from the Aggregate Net Loss to yield the "Final Aggregate Net Loss."

- F. The Settlement Administrator shall then calculate for each Member who was not a De Minimis Loss Member (i) his or her "Final Fractional Share" by dividing the Net Loss of each Member who was not a De Minimis Loss Member by the Final Aggregate Net Loss, and (ii) his or her "Final Dollar Recovery" by multiplying the Net Proceeds by each such Members' Final Fractional Share.
- G. Any portion of the Net Proceeds due to Unlocatable Members shall be administered in accordance with the Plans' procedures regarding unlocatable participants.
- H. In light of the manner in which the Plan data are kept and the efficacy with which the Plan data can be used to locate Members and to allocate each Member's Final Dollar Recovery, it may be appropriate to modify some of the features of these calculations. Such modifications shall be acceptable as long as each Member is allocated his or her share of the Net Proceeds based approximately on the decline in the value of units of the Xerox Stock Fund that he or she held over the Allocation Period as a proportion of the decline in the value of units of the Xerox Stock Fund held by all Members.

III. Distribution of the Allocated Amounts

- A. As soon as administratively feasible after calculating each Member's Final Dollar Recovery pursuant to section II, *supra*, Co-Lead Counsel shall direct the Financial Institution to disburse the *Net Proceeds* to the master trust for the *Plans* for allocation to the Members in accordance with this Plan of Allocation; such disbursement shall be in accordance with wire transfer instructions to be provided by Xerox or the Trustee.
 - B. The Final Dollar Recoveries of the Current Plan Participants shall be allocated to

each Member's Plan account in their respective Plan and allocated to the Income Fund investment option, or the successor thereto, and administered thereafter for all purposes in accordance with that Plan.

C. The aggregate Final Dollar Recoveries of the Former Plan Participants in the respective Plans shall be invested in a money market fund that invests primarily in United States treasury instruments, pending allocation to the Former Plan Participants. The deposited amount, plus interest, shall then be allocated as soon as is practical to the Plan accounts established for the Former Plan Participants in their respective Plans and allocated to the Income Fund investment option, or the successor thereto, and administered thereafter for all purposes in accordance with the Plan.

EXHIBIT C TO CLASS ACTION SETTLEMENT AGREEMENT

(PLAN OF ALLOCATION)

PLAN OF ALLOCATION

I. Definitions

- A. All capitalized terms shall have the same meaning as they are given in the Settlement Agreement dated ______ ("Settlement Agreement"), unless specifically set forth otherwise in this Plan of Allocation. The following definitions shall apply to the Plan of Allocation methodology:
- 1. **"Plan Account"** means the account maintained on behalf of a participant in either of the Plans.
- 2. "Member" shall mean a member of the Settlement Class who is or was a Participant and whose Plan Account held any units in the Xerox Stock Fund during the Allocation Period.
- 3. "Current Plan Participant" shall mean: (i) a Member who as of the Reference Date has not received a complete distribution from a Plan; and (ii) a beneficiary or beneficiaries of such a Member.
- 4. "Allocation Period" means, for each Member, the period through and including (a) the earlier of (i) January 2, 1998, or (ii) such date from May 12, 1997 through January 1, 1998, as to which such Member provides to the Settlement Administrator "Adequate 1997 Documentation", and (b) June 28, 2002.
- 5. "Adequate 1997 Documentation" means documentation provided by a Member to the Plan Administrator no later than sixty (60) days after the commencement of the Fairness Hearing, which documentation (i) was created by either of the Plans or their record-keeper or Trustee, or was otherwise contemporaneously created, (ii) shows the acquisition, holding or disposition of units in the Xerox Stock Fund by in such Member's Plan Account

during the period from May 12, 1997 through January 1, 1998, and (iii) is determined by the Settlement Administrator in its discretion to provide sufficiently reliable information for the Settlement Administrator to be able to use it in calculating such Member's Net Loss in accordance with the provisions of Section II of this Plan of Allocation.

- 6. **"Former Plan Participant"** shall mean (i) a Member who, as of the Reference Date, is not a Current Plan Participant; and (ii) a beneficiary or beneficiaries of such a Member.
- 7. "**Xerox Stock**" means the common stock of Xerox Corporation ("Xerox") traded on the New York Stock Exchange under the symbol XRX.
- 8. **"Participant**" means a person who was a participant within the meaning of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), in a Plan during the Allocation Period.
- 9. **"Unlocatable Member"** shall mean a Member who, despite reasonable efforts, cannot be located by the Plan Administrator within a reasonable time.
- 10. **"Reference Date"** means the date after the Judgment becomes final, established by the Settlement Administrator for determination of whether a Participant is a Current Plan Participant.

II. Calculation of Allocation

A. As soon as administratively feasible after the Settlement becomes final in accordance with the terms of the Settlement Agreement, the Settlement Administrator shall calculate the share of the Net Proceeds for each Member according to the following methodology:

2

- B. For each Member, the Settlement Administrator shall determine a net loss ("Net Loss") as follows:
 - 1. Net Loss =(A+B)-(C+D), where, for each Member's Plan Account:
- 2. A = the total dollar value of his or her Plan Account balance in the Xerox Stock Fund at the beginning of the Allocation Period;
- 3. B = the dollar value, if any, of all subsequent investments of any nature whatsoever of such Plan Account in units in the Xerox Stock Fund during the Allocation Period. valued as of the time of each such investment as reflected in the records of a Plan;
- C = the dollar amount of all dispositions of any nature whatsoever of units 4. in the Xerox Stock Fund in the Plan Account, if any, during the Allocation Period, valued as of the time of the disposition, as reflected in the records of a Plan; and
- D = the dollar value of the Plan Account invested in the Xerox Stock Fund 5. at the end of the Allocation Period.
- C. The Settlement Administrator will then total the Net Losses of all Members as calculated in Section II.B to yield the "Aggregate Net Loss," and shall then divide each Member's Net Loss by the Aggregate Net Loss to yield each Member's "Preliminary Fractional Share."
- D. The Settlement Administrator shall then calculate for each Member his or her "Preliminary Dollar Recovery" of the Net Proceeds by multiplying the Member's Preliminary Fractional Share by the Net Proceeds.
- E. The Settlement Administrator shall then identify all Members whose Preliminary Dollar Recovery is less than fifteen dollars (\$15.00) (the "De Minimis Amount"). All such Members' losses shall be deemed to be zero and no allocation will be made to such Members

3

(the "De Minimis Loss Members"). The aggregate Net Losses of the De Minimis Loss Members shall be subtracted from the Aggregate Net Loss to yield the "Final Aggregate Net Loss."

- F. The Settlement Administrator shall then calculate for each Member who was not a De Minimis Loss Member (i) his or her "Final Fractional Share" by dividing the Net Loss of each Member who was not a De Minimis Loss Member by the Final Aggregate Net Loss, and (ii) his or her "Final Dollar Recovery" by multiplying the Net Proceeds by each such Members' Final Fractional Share.
- G. Any portion of the Net Proceeds due to Unlocatable Members shall be administered in accordance with the Plans' procedures regarding unlocatable participants.
- H. In light of the manner in which the Plan data are kept and the efficacy with which the Plan data can be used to locate Members and to allocate each Member's Final Dollar Recovery, it may be appropriate to modify some of the features of these calculations. Such modifications shall be acceptable as long as each Member is allocated his or her share of the Net Proceeds based approximately on the decline in the value of units of the Xerox Stock Fund that he or she held over the Allocation Period as a proportion of the decline in the value of units of the Xerox Stock Fund held by all Members.

III. Distribution of the Allocated Amounts

- A. As soon as administratively feasible after calculating each Member's Final Dollar Recovery pursuant to section II, *supra*, Co-Lead Counsel shall direct the Financial Institution to disburse the *Net Proceeds* to the master trust for the *Plans* for allocation to the Members in accordance with this Plan of Allocation; such disbursement shall be in accordance with wire transfer instructions to be provided by Xerox or the Trustee.
 - B. The Final Dollar Recoveries of the Current Plan Participants shall be allocated to

each Member's Plan account in their respective Plan and allocated to the Income Fund investment option, or the successor thereto, and administered thereafter for all purposes in accordance with that Plan.

C. The aggregate Final Dollar Recoveries of the Former Plan Participants in the respective Plans shall be invested in a money market fund that invests primarily in United States treasury instruments, pending allocation to the Former Plan Participants. The deposited amount, plus interest, shall then be allocated as soon as is practical to the Plan accounts established for the Former Plan Participants in their respective Plans and allocated to the Income Fund investment option, or the successor thereto, and administered thereafter for all purposes in accordance with the Plan.

Other Documents

3:02-cv-01138-AWT Patti, et al v. Xerox Corp, et al

DFM, EFILE, LEAD, MOTREF, STAYED

U.S. District Court

United States District Court for the District of Connecticut

Notice of Electronic Filing

The following transaction was entered by Leland, Elizabeth on 1/26/2009 at 6:18 PM EST and filed on 1/26/2009

Case Name: Patti, et al v. Xerox Corp, et al

Case Number: 3:02-cv-1138

Filer: Thomas Patti

Cheryl L. Wright David Alliet Linda Willis

Document Number: 340

Docket Text:

SETTLEMENT AGREEMENT Class Action Settlement Agreement by Linda Willis, David Alliet, Thomas Patti, Cheryl L. Wright. (Attachments: # (1) Ex. A to Class Action Settlement Agreement - Order Preliminarily Approving Settlement and Confirming Final Settlement Hearing, # (2) Ex. B to Class Action Settlement Agreement - Order and Final Judgment, # (3) Ex. C. to Class Action Settlement Agreement - Plan of Allocation) (Leland, Elizabeth)

3:02-cv-1138 Notice has been electronically mailed to:

Rosemary Q. Barry rqbarry@daypitney.com, sprice@daypitney.com

Elizabeth K. Canizares elizabeth.canizares@wilmerhale.com

Joy P. Clairmont jclairmont@bm.net

Ellen M. Doyle edoyle@stemberfeinstein.com, filings@stemberfeinstein.com, jhurt@stemberfeinstein.com, pewing@stemberfeinstein.com, spincus@stemberfeinstein.com

Shay Dvoretzky sdvoretzky@jonesday.com

Terence J. Gallagher, III tigallagher@daypitney.com, eecaie@daypitney.com

Thomas D. Goldberg tdgoldberg@daypitney.com, vaboswell@daypitney.com

Gary A. Gotto GGotto@kellerrohrback.com

Benjamin H. Green bgreen@pastoreosterberg.com

Blake T. Hannafan bth@hannafanlaw.com

Michael T. Hannafan mth@hannafanlaw.com

James A. A. Kirk jkirk@jonesday.com

Rebecca B. Lamont rlamont@grmattorneys.com

Elizabeth A. Leland bleland@kellerrohrback.com, chopkins@kellerrohrback.com, dwilcher@kellerrohrback.com

Derek W. Loeser dloeser@kellerrohrback.com

Arthur J. Margulies ajmargulies@jonesday.com, kpollak@JonesDay.com

John F. McKenna johnm@grmattorneys.com, bripa@grmattorneys.com, jmckenna@grmattorneys.com

James Brian McTigue bmctigue@mctiguelaw.com, dbond@mctiguelaw.com, jgorka@mctiguelaw.com, jstrouf@mctiguelaw.com

Felipe D. Mendoza felipe.mendoza@wilmerhale.com

Evan Miller em iller@jonesday.com

Kevin R. Noble krnoble@jonesday.com

Sara Pikofsky spikofsky@jonesday.com

Joanne Rapuano jrapuano@dreierllp.com

Erin M. Riley eriley@kellerrohrback.com

Steven J. Sacher sjsacher@jonesday.com

Lynn Lincoln Sarko lsarko@kellerrohrback.com, cengle@kellerrohrback.com

Sherrie R. Savett ssavett@bm.net

Aviva W. Sisitsky aswarter@jonesday.com, dpjacobson@jonesday.com, elgoodman@jonesday.com

Frances Codd Slusarz fslusarz@pastoreosterberg.com, fslusarz@comcast.net

Jennifer H. Strouf jstrouf@mctiguelaw.com

Jayant W. Tambe jtambe@jonesday.com

John A. Valentine john.valentine@wilmerhale.com

Charles R. Watkins Cwatkins@futtermanhoward.com

John R. Wylie jwylie@futtermanhoward.com

Patrick P. de Gravelles pdegravelles@mctiguelaw.com

3:02-cv-1138 Notice has been delivered by other means to:

Thomas Patti

Alexander Perkins Keller, Rohrback 1201 Third Ave. Suite 3200 Seattle, WA 98101

The following document(s) are associated with this transaction:

Document description: Main Document

Original filename:n/a

Electronic document Stamp:

[STAMP dcecfStamp_ID=1034868047 [Date=1/26/2009] [FileNumber=1849584-0] [46a8a9f363ef7a7615e8ccf87260bf09e11de505996e3085793e2ec811b4b43f277 17190914109a53f259205754e7a59542c58ed2c93bda354462471da0bf8df]]

Document description: Ex. A to Class Action Settlement Agreement - Order Preliminarily Approving Settlement and Confirming Final Settlement Hearing

Original filename:n/a

Electronic document Stamp:

[STAMP dcecfStamp_ID=1034868047 [Date=1/26/2009] [FileNumber=1849584-1] [83f481e173f23a8ce1b31630b5c07282442d21330c30d870d08d5e079d3bf6fa7e2 7cb49918aa7627a9be91665f2c05d02fbc0a16c2228ec2fa7bb1800cb86d1]]

Document description: Ex. B to Class Action Settlement Agreement - Order and Final Judgment

Original filename:n/a

Electronic document Stamp:

[STAMP dcecfStamp_ID=1034868047 [Date=1/26/2009] [FileNumber=1849584-2] [6de07c1dfd264d12ead048cdf8f8b035b81b153396338d91c360fdfd95d2d91d1f7 c9f83e4a6ee43033a1e0090abe50d9d4877c221844b59981f1969d404e7e8]]

Document description: Ex. C. to Class Action Settlement Agreement - Plan of Allocation

Original filename:n/a

Electronic document Stamp:

[STAMP dcecfStamp_ID=1034868047 [Date=1/26/2009] [FileNumber=1849584-3] [2348d56f98d0beea846d39aeb48154a21e5dd84dfb82b430599563fbfcd28ec8ac5 7e64da5a2eaaf0704e188a3141485441fcd2e6eb7ba819823a019b8a57280]]